ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Van den Berg, Gerard J.; Vikström, Johan

Working Paper Long-run effects of dynamically assigned treatments: A new methodology and an evaluation of training effects on earnings

Working Paper, No. 2019:18

Provided in Cooperation with:

IFAU - Institute for Evaluation of Labour Market and Education Policy, Uppsala

Suggested Citation: Van den Berg, Gerard J.; Vikström, Johan (2019) : Long-run effects of dynamically assigned treatments: A new methodology and an evaluation of training effects on earnings, Working Paper, No. 2019:18, Institute for Evaluation of Labour Market and Education Policy (IFAU), Uppsala

This Version is available at: https://hdl.handle.net/10419/227838

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Long-run effects of dynamically assigned treatments:

a new methodology and an evaluation of training effects on earnings

Gerard J. van den Berg Johan Vikström



Long-Run Effects of Dynamically Assigned Treatments: a New Methodology and an Evaluation of Training Effects on Earnings^a

by

Gerard J. van den Berg^b, and Johan Vikström^c

July, 2019

Abstract

We propose and implement a new method to estimate treatment effects in settings where individuals need to be in a certain state (e.g. unemployment) to be eligible for a treatment, treatments may commence at different points in time, and the outcome of interest is realized after the individual left the initial state. An example concerns the effect of training on earnings in subsequent employment. Any evaluation needs to take into account that some of those who are not trained at a certain time in unemployment will leave unemployment before training while others will be trained later. We are interested in effects of the treatment at a certain elapsed duration compared to "no treatment at any subsequent duration". We prove identification under unconfoundedness and propose inverse probability weighting estimators. A key feature is that weights given to outcome observations of non-treated depend on the remaining time in the initial state. We study earnings effects of WIA training in the US and long-run effects of a training program for unemployed workers in Sweden. Estimates are positive and sizeable, exceeding those obtained by using common static methods, and suggesting a reappraisal of training.

Keywords: Treatment effects; dynamic treatment evaluation; program evaluation; duration analysis; matching; unemployment; employment.

^aWe thank Bart Cockx, Bruno Crépon, Bernd Fitzenberger, Anders Forslund, Peter Fredriksson, Martin Lundin, Jeffrey Smith, participants at seminars at Humboldt University Berlin, Ghent University, IFAU, Uppsala Center for Labor Studies, the Swedish Ministry of Finance, the Swedish Public Employment Service, and conference participants at IFAU/IZA, SOLE/EALE, ESWC, RES, Cafe Aarhus and COMPIE conferences, for useful suggestions. Vikström acknowledges support from FORTE.

^bUniversity of Bristol, IFAU, IZA, ZEW, CEPR and CESifo

[°]IFAU and Uppsala Center For Labor Studies, Uppsala University, E-mail: johan.vikstrom@ifau.uu.se

Table of contents

1	Introduction	3	
2	Model	8	
3	The evaluation problem	0	
4 4.1 4.2	Identification1The gist of the approach in a simple setting1Identification results1	3 3 5	
5	Estimation	0	
6 6.1 6.2 6.3	Extensions2Short-run and long-run outcomes2Time-varying covariates2Right-censored durations2	3 3 7 8	
7 7.1 7.2 7.3 7.4	WIA training in the US3Background3The dynamic evaluation setting3Data and estimation details3Estimation results3	1 1 3 7	
8 8.1 8.2	Long-run effects of Swedish training programs 33 Background and data 33 Estimation results 4	8 8 1	
9	Conclusions	3	
Refe	References		
App	Appendix A. Proofs and derivations 50		
App	Appendix B. Monte Carlo simulation		

1 Introduction

By now, an extensive toolkit has been developed for treatment evaluations based on unconfoundedness assumptions (i.e., absence of unobserved confounders). This includes the construction of efficient and robust estimators, balancing tests and the imposition of common support restrictions (e.g., Imbens and Wooldridge, 2009). Much of this literature considers a static evaluation problem, where the comparison is between being treated and not being treated. Training programs and other active labor market policy (ALMP) programs have often been evaluated using such methods, especially if rich register and/or survey data are available. However, the static framework is of limited value for the evaluation of such programs. To see this, consider, more generally, a setting where eligibility for a treatment requires the individual to be in a certain state. Within this initial state, the date of entry into the treatment may vary, for example because of capacity constraints on the number of treatment slots. The duration of being in the initial state may vary as well, for example because of market frictions. Some of those who do not enter the treatment at a certain point in time will therefore leave the initial state before they are ever treated, while others will be treated before they leave. In such a setting, the evaluation needs to take the dynamic nature of the treatment assignment into account, and this complicates the selection of a proper control group. The literature contains several contributions for the case where the length of stay in the initial state is the outcome of interest (see the literature discussion below).

In this paper, we consider outcome measures that are realized after the individual has left the initial state. Such outcomes are often deemed interesting. Compared to outcomes capturing the length of stay in the initial state, they are less sensitive to short-run institutional features and market imperfections. A case in point is the evaluation of post-unemployment earnings effects of participation in a training program for unemployed individuals. In the short run, locking-in effects of program participation may dominate the effect on productivity, whereas in the longer run, outcomes may better reflect the best attainable earnings levels, and hence the estimated effect may better reflect the effect on productivity or on human capital. In addition to this, longer-run outcomes may be the key

drivers of the present values of the various treatment regimes, and hence they are likely to play an important role in cost-benefits analyses.¹

Long-run data on individuals who are treated at the elapsed duration *t* in the eligibility state are informative on the average potential outcome under treatment, whereas *only* the individuals who leave the initial state before treatment provide information on no-treatment potential outcomes. Ultimately, the realized treatment status depends on the realized time in the initial state. The challenge that we face here is that the non-treated group cannot be straightforwardly used to construct an average counterfactual outcome for the treated.

This may be illustrated by the example with training during unemployment and its effect on long-run earnings.² With a sufficiently long time horizon, the long-run earnings of the treated at t are directly observable. Under unconfoundedness among survivors in unemployment at t we can adjust for differences between treated and those who are not treated at t. However, in subsequent time periods some of the previously non-treated will receive the treatment before they leave unemployment as well. Individuals with a small probability of receiving the treatment in subsequent periods are more likely to remain non-treated. Moreover, for a given period-by-period treatment probability, the individuals with a high rate of leaving unemployment are more likely to remain non-treated. Both features (small probability of treatment in subsequent periods and high exit rate) may influence long-run earnings. To infer the causal effect of training, these features need to be controlled for.

In this paper, we develop a novel approach to resolve this, based on unconfoundedness (or conditional independence) assumptions. We define average treatment effects on long-

¹Many studies of European training programs focus on transitions from unemployment to employment as the key outcome. One reason for this is the concern about relatively long unemployment durations. Moreover, with long durations, post-unemployment outcomes have often only been observable for a modest subset of the sample, leading to concerns about selective observability (Ham and LaLonde, 1996). Recently, the interest in long-run outcomes has been increasing (see e.g., the survey of Crépon and Van den Berg, 2016). Studies of US training programs have always focused more on earnings outcome variables.

²Besides training and post-unemployment earnings, examples can be constructed where the treatment is the usage of a particular medication and the eligibility period is the spell of a certain illness during which the medication is prescribed. Alternatively, one may consider intergenerational settings where the treatment is an event in the life of a child (e.g., the achievement of a certain level of education) and the outcome relates to features of the parents' life after the decision not to have additional children. Some individuals may never be treated in the eligibility state, as they will be childless at high ages and are unlikely to still obtain a child.

run outcomes among those who are treated at a given elapsed time in the initial state, and we aggregate this into average treatment effects for the full program. We introduce and implement inverse probability weighting (IPW) estimators for the various average effects. Besides unconfoundedness we do not impose any structure on the assignment process. We make the usual no-anticipation assumption (see e.g., Abbring and van den Berg, 2003) on the distribution of potential outcomes. To identify the counterfactual, we *only* use outcome observations of those who are not treated before leaving unemployment. As it turns out, the weights given to those observations do not only depend on observed confounders but also implicitly on the time spent in unemployment. Such weights establish a balance between the treated population at t and the control group consisting of the unemployed who are non-treated throughout the entire unemployment spell. We provide identification results and investigate the theoretical properties of our estimator in detail. Simulation results confirm that the IPW estimator is asymptotically unbiased and that the proposed bootstrap inference works well.

The baseline identification and estimation results are for a setting with selection based on time-invariant observed characteristics (covariates), absence of censoring of the durations in the initial state, and with only one long-run outcome realized after the individuals left the initial state. We then generalize the results to several different settings. This includes allowing for selection on *time-varying* covariates. For instance, in evaluations of training programs, the selection to the training may depend on characteristics that change during the unemployment spell. Another extension is to allow for right-censored durations which are common in many applications, for instance, due to drop-out from the study or a limited follow-up period. We also consider joint effects on both short-run and long-run outcomes. For training programs for unemployed workers, the short-run outcomes may be earnings a couple of months after the start of the training.

These methodological contributions are related to several strands of literature. A substantial body of work faces dynamic treatment assignment, either if the outcome of interest is the sojourn time in the initial eligibility state, such as the average unemployment duration after treatment,³ or if the outcome of interest is realized partly or fully after unemployment. For post-unemployment outcomes, Lechner (1999, 2002), Wunsch and Lechner (2008) and Lechner et al. (2011) developed the "hypothetical treatment durations" approach, where hypothetical treatment dates are generated for each non-treated individual, and the actual and hypothetical treatment dates are used as covariates in the propensity score.⁴ Another approach compares "treatment now versus treatment later", as introduced by Sianesi (2004), with applications in e.g. Sianesi (2008), Fitzenberger et al. (2008) and Biewen et al. (2014). For a given pre-treatment duration, it uses individuals who are not treated at this duration but possibly treated later on, as a control group. Note that this redefines the treatment effect.

As mentioned above, if the post-unemployment outcome is realized strictly after completion of the eligibility period, then, depending on the circumstances, the outcome may not be observed for a substantial fraction of the full sample. Lee (2009) and Lechner and Melly (2010) propose conditions under which average effects on the outcome can be bounded. We do not impose such conditions, and we mainly focus on long-run outcomes that are observed for every sample member. This holds by approximation for outcomes that are observable at a sufficiently long period of time after entry into the eligibility state. A number of other studies use random-effects models to jointly estimate duration effects and post-unemployment effects of exposure to labor market policy programs, allowing for selection on unobservables. This approach effectively extends the Timing-of-Events (ToE) approach by Abbring and Van den Berg (2003) developed for duration effects, by adding model equations for post-unemployment outcomes. In contrast, our approach does not impose model equations and does not restrict the heterogeneity of treatment effects.

Our paper contains two full-blown empirical applications; one for the US and one for

³In a recent study, Vikström (2017) proposes an IPW estimator of average effects on the survival probability of being in the state of interest under unconfoundedness. That estimator is not designed with an eye on long-run outcomes and it essentially uses all not-yet treated in a period as the control group for treatment in that period. Note that in our case the control group only consists of non-treated, i.e. those who leave the state without having been treated. For that reason, the weights also differ. See Crépon et al. (2009) for a similar estimator based on blocking methods for the propensity score, and Muller et al. (2017) for a comparison of existing dynamic evaluation approaches for duration outcomes.

⁴As Lechner (1999) writes, many ways can be defined to assign values to hypothetical treatment dates, and although they are intuitively plausible, it appears difficult to derive properties of the various approaches based on hypothetical treatment dates analytically.

Sweden. First, we examine the WIA training program in the US (see e.g., Andersson et al., 2016), which includes counselling and training services. We follow several previous studies and estimate the effect of training relative to other WIA services, i.e. the effect of training conditional on participation in WIA. This setting is similar to the example with training in unemployment. The WIA training program has an eligibility state (being registered at a WIA job center) within which individuals can enroll in WIA training. Assignment to WIA training is a dynamic process in which training starts after some elapsed duration in WIA. Thus, some of the currently non-trained will receive training before they leave WIA.

The results show that WIA training has positive effects on earnings after exit from WIA. Comparisons with a standard static evaluation approach reveal that the static approach underestimates the WIA training effect by roughly 20% (the difference is significant). This is because the control group in a static approach includes too many participants who only remain in WIA for a short period, leading to downward bias since this, most likely, is a group with more favorable characteristics. The new dynamic approach in this paper corrects for this selection problem.

The second empirical application considers long-run earnings effects of the flagship training program for unemployed workers in Sweden. The main purpose of this program is to improve the skills and thereby to enhance reemployment chances. Starting with Regnér (2002), studies have concluded that the program did not raise post-unemployment earnings (see also the survey in Forslund and Vikström, 2011). This has been a reason for policy makers to cut down on training program volumes.⁵ Our data allow us to follow each unemployed individual in our sample for more than ten years after entry into unemployment, and our method allows us to properly correct for dynamic assignment to training. We find that in the short run the program decreases annual labor earnings. This reflects lock-in effects, since some of these earnings are received while still in unemployment. However, in the medium and long run, the estimated earnings effects of the program are persistently significantly positive, leading to positive effects on cumulative

⁵More recent evaluations from the Swedish Public Employment Office with negative employment effects of training have also contributed to the decreased program volumes (e.g., PES, 2016).

earnings. We conclude that previous evaluations need to be revised in the light of these substantial positive earnings effects in the long run.

The outline of the paper is as follows. Section 2 presents the causal framework. Section 3 lays out the evaluation problem, and Section 4 contains the main identification results. Section 5 discusses estimation and statistical inference. In Section 6, we extend the results to different settings. Section 7 contains the US application and Section 8 reports the results from the Swedish application. Section 9 concludes.

2 Model

Consider an individual entering a state in which (s)he is eligible for a treatment. This state is called the initial state or eligibility state. The treatment may take place at some point in time while in the initial state but the individual may also leave this state at some point in time without having been treated. We are interested in the effect of the treatment on long-run outcomes. Initially, we consider the case when the long-run outcome is realized after individuals left the initial state. In Subsection 6.1 we allow the observation period for the long-run to overlap with the initial spell.

Let the time clock start at the moment of entry into the initial state. We denote the duration spent in the initial state by T_u and the duration until treatment by T_s . Note that in this framework T_s is a latent variable as it can be right-censored by T_u . The potential time in the initial state if the individual is assigned to be treated at t_s is denoted by $T_u(t_s)$. The long-run outcome is denoted by Y, and its corresponding potential outcomes by $Y(t_s)$. Thus, $T_u(\infty)$ and $Y(\infty)$ capture the potential duration and the potential long-run outcome if the individual is assigned to be "never treated". In practice, infinity may be replaced by some upper bound on the eligibility interval length. (In the next subsection we discuss the role of $T_u(t_s)$ in the long-run effects on $Y(t_s)$.)

We take time to be discrete. We assume that, within each time period t with $T_s \ge t$, the binary event governing whether $T_s = t$ vs. $T_s > t$ is realized before the binary event governing whether $T_u = t$ vs. $T_u > t$. This means that we do not impose a one-period delay in the effect of t_s on T_u . We adopt the "no anticipation" assumption (Abbring and Van den Berg, 2003), ruling out that the timing of a future treatment affects the current probability of leaving the initial state. Specifically, for each individual,

Assumption 1 (No anticipation in exit out of the eligibility state)

$$P(T_u(t') = t) = P(T_u(t'') = t), \quad \forall t < \min(t', t'')$$

This implies that $P(T_u(t') = t) = P(T_u(\infty) = t), \forall t < t'.$

We also require a no-anticipation assumption regarding $Y(t_s)$. Specifically,

Assumption 2 (No anticipation in long-run outcomes)

$$Y(t') = Y(t''), \quad \forall T_u(t') = T_u(t'') < \min(t',t'').$$

This rules out that the potential outcomes $Y(t_s)$ are affected by behavior before t_s that is driven by the future treatment date t_s , even if this behavior is of no consequence for the moment of leaving the eligibility state before t_s . For example, what is ruled out is the following: while in the eligibility state, the individual acts upon private knowledge of his future moment of treatment in such a way that his long-run outcome is affected but his time in the eligibility state is not. (Note that *if* the probability of leaving would be affected by such behavior before t_s then Assumption 1 is violated.) Assumption 2 only considers $T_u(t') = T_u(t'')$ because if $T_u(t') \neq T_u(t'')$ then a difference in the long-run outcomes Y(t') and Y(t'') may be ascribed to an effect of T_u on Y. Together, Assumptions 1 and 2 ensure that individual outcomes do not vary with the moment of treatment as long as the treatment is not realized. In the remainder of the paper we therefore often simply refer to Assumptions 1 and 2 jointly as "the" no-anticipation assumptions.

Recall that in the model structure in this section, Y is always realized after T_u . Later on, when we allow the observation period for the long-run to overlap the initial spell, we need to adopt an alternative no-anticipation assumption for $Y(t_s)$.

In the absence of right-censoring of T_u , and by invoking Assumptions 1 and 2, we can express the actual long-run outcome *Y* as follows,

$$Y = \sum_{t=1}^{T_u} [\mathbf{I}(T_s = t)Y(t)] + \mathbf{I}(T_s > T_u)Y(\infty),$$
(1)

where I() is the indicator function. This is the observational rule. The first part of the right-hand side of (1) states that if the individual is treated before leaving the initial state, then we observe the long-run potential outcome corresponding to the actually observed time to treatment. The second part states that if an individual exits the initial state without having been treated then the observed outcome is the outcome corresponding to the assignment "never treated".

3 The evaluation problem

A possible object of interest is the average treatment effect of t_s on Y among those who are actually treated at t_s . This is referred to as ATET (t_s) :

$$ATET(t_s) = E(Y(t_s) - Y(\infty) \mid T_s = t_s, T_u(t_s) \ge t_s).$$
(2)

This contrasts the treatment at t_s with being given the assignment to be "never treated". By aggregating this over the distribution of T_s we obtain the average effect of the program in comparison to a setting where the program is absent (ignoring equilibrium effects).

An alternative object of interest is the average treatment effect of t_s on Y among all those who, if they were assigned to t_s , would still be in the initial state at that time t_s . Contrary to ATET (t_s) , this object does not depend on the actual assignment mechanism. We refer to it as ATE (t_s) ,

$$ATE(t_s) = E(Y(t_s) - Y(\infty) \mid T_u(t_s) \ge t_s).$$
(3)

In general, these average causal effects of t_s on the long-run outcome $Y(t_s)$ may include (1) a direct effect and (2) an indirect that runs by way of the time $T_u(t_s)$ spent in the initial eligibility state. For example, with training programs for unemployed workers, (1) may capture that the training program improves the participants' human capital and the ensuing labor earnings, while (2) may capture that program participation speeds up

exit out of unemployment and the resulting unemployment duration is used by employers as a signal of worker quality and hence as a determinant of earnings. In the long-run, as employers gather more and more information on worker quality, (or in a flexible labor market with full information,) indirect effects are likely to lose importance, so that the over-all effect primarily represents the direct effect. However, depending on the application at hand, it cannot be ruled out that the over-all effect includes an indirect effect, especially if the treatment is only targeted at the reduction of the time in the eligibility state.

In this paper, we do not attempt to separate the direct effect from the indirect effect. In different words, we examine the over-all comprehensive effect on Y regardless of the pathway. This is why the potential outcome notation $Y(t_s)$ does not explicate a causal effect of t_u on Y. In the sequel, in line with the "matching" evaluation literature on which we build, we mainly focus attention on ATET (t_s) .

We denote the set of observed individual characteristics by *X*. Concerning the assignment process, we assume sequential unconfoundedness. For presentation reasons we will initially explain our identification results under the assumption that unconfoundedness holds conditional on *time-invariant* covariates. Subsection 6.2 generalizes the results allowing for selection on *time-varying* covariates.

Formally, let the binary indicator $P_t := 1$ iff $T_s = t$.

Assumption 3 (Assignment) For all t,

$$P_t \bot \{Y(t_s)\}_t^{\infty} \mid X, T_s \ge t, T_u(\infty) \ge t.$$

Here, the joint distribution of $\{Y(t_s)\}_t^{\infty}$ is taken to cover all values $t_s = t, t + 1, ..., \infty$. In fact, inference on ATET (t_s) only requires P_t to be conditionally independent of the potential outcome $Y(\infty)$ if non-treated, while the full Assumption 3 is needed for ATE (t_s) .

This assumption echoes the usual CIA unconfoundedness assumptions in the evaluation literature based on matching methods, including the references cited above. Somewhat informally, it rules out that there are systematic unobserved characteristics that drive both the treatment assignment and the long-run outcomes. In the case of training programs for unemployed workers, the argument has been made that caseworkers determine the assignment and that the data set used in the analysis include the information used by the caseworkers when deciding on the assignment. The assumption is also more likely to hold in cases with detailed administrative data and/or survey data that capture individual motivation, ability, and personality traits that may affect treatment assignments as well as outcomes.

As usual in non-experimental evaluations, an overlap condition or common support condition is needed. In the basic evaluation setting, this amounts to assuming that for all t and X we have that $Pr(P_t = 1 | T_s \ge t, T_u \ge t, X) < 1$. In addition, the usual SUTVA assumption needs to hold, ruling out various types of interference between the units in the sample (see e.g. Wooldridge, 2010). In the sequel, for expositional convenience, we tacitly assume common support and SUTVA wherever applicable, and we do not spell out the corresponding requirements in the derivations.

To explain the evaluation problem, consider first the first component of the expression for $ATET(t_s)$. From a random sample of observed outcomes *Y* of those treated at time T_s , this component is straightforwardly identified:

$$E(Y(t_s)|T_s = t_s, T_u(t_s) \ge t_s) = E(Y|T_s = t_s, T_u \ge t_s).$$
(4)

The evaluation problem concerns the second component of $ATET(t_s)$, that is, the counterfactual outcome $E(Y(\infty)|T_s = t_s, T_u(t_s) \ge t_s)$. The observational rule in (1) illustrates the main issues. First, it is clear that only non-treated individuals, i.e. individuals with $T_s > T_u$, can be informative on the counterfactual outcome under never treatment. Individuals who are not treated at t_s , but are treated later are not informative on the counterfactual outcome under never treatment. After all, since they are ultimately treated, their observed response, Y, corresponds to a potential outcome under a treatment.

Second, the potential control group of non-treated, defined by $T_s > T_u$, in general, constitutes a selective sample. Indicators like $I[T_s > T_u | T_s = t_s, T_u \ge t_s]$ depend on the survival time T_u (i.e., on an intermediate outcome which may affect the outcome of interest), since the probability of treatment enrollment by construction increases with the time in

the initial state. Thus, for a given period-by-period treatment probability, individuals with longer unemployment durations are less likely to remain non-treated. This arises purely because of the dynamic nature of the treatment assignment. To phrase this differently, individuals with relatively short durations after a treatment at t_s are overrepresented in the group of non-treated. Thus, even if treatment assignment is unconfounded, treatment assignments in subsequent time periods cause the group of non-treated to be systematically different.

4 Identification

4.1 The gist of the approach in a simple setting

This subsection describes the gist of our method in a simplified setting with two periods in which the treatment can take place whereas no treatments can occur thereafter. We consider the effect of treatment in the first period, $\text{ATET}(1) = E(Y(1) - Y(\infty) | T_s = 1)$. (Note that for the effect of a treatment in the first period there is no need to condition on survival.) Subsequent sections generalize this setting.

As a starting point, note that $E(Y(1)|T_s = 1) = E(Y|T_s = 1)$ follows directly from the observational rule. The identification challenge is to show that the counterfactual $E(Y(\infty)|T_s = 1)$ is identified under our assumptions. If we condition on X then, by the assignment-assumption (Assumption 3), the treated and non-treated are comparable:

$$E(Y(\infty)|T_s = 1, X) = E(Y(\infty)|T_s > 1, X).$$
(5)

By the law of iterated expectations

$$E(Y(\infty)|T_{s} > 1, X) =$$

$$Pr(T_{u} = 1|T_{s} > 1, X)E(Y(\infty)|T_{s} > 1, T_{u} = 1, X) +$$

$$Pr(T_{u} > 1|T_{s} > 1, X)E(Y(\infty)|T_{s} > 1, T_{u} > 1, X).$$
(6)

That is, the counterfactual outcome under never treatment is decomposed into average outcomes for individuals with $T_u = 1$ and with $T_u > 1$. Note that probabilities $Pr(T_u =$

 $1|T_s > 1, X)$ and $Pr(T_u > 1|T_s > 1, X)$ are observed.

Next, the group with $T_u = 1$ in (6) consists of non-treated individuals who exit directly in period 1. For this group, the observational rule in (1) and no-anticipation (Assumptions 1 and 2) give

$$E(Y(\infty)|T_s > 1, T_u = 1, X) = E(Y|T_s > 1, T_u = 1, X).$$
(7)

This follows since for non-treated individuals with $T_s > T_u$, the observed long-run outcome *Y* equals the non-treated long-run potential outcome $Y(\infty)$.

For the group with $T_u > 1$, in (6), i.e., those who survive at least one additional time period, we have by the assignment-assumption (Assumption 3):

$$E(Y(\infty)|T_s > 1, T_u > 1, X) = E(Y(\infty)|T_s > 2, T_u > 1, X),$$
(8)

i.e. among the non-treated survivors, conditional on *X*, those who become treated in period 2 are comparable to those who remain non-treated. Next, since there are no treatments after period 2 and under no-anticipation, those with $T_s > 2$ remain non-treated, so that for this group the observed long-run outcome *Y* equals the non-treated long-run potential outcome, $Y(\infty)$:

$$E(Y(\infty)|T_s > 2, T_u > 1, X) = E(Y|T_s > 2, T_u > 1, X).$$
(9)

Then, by (5)-(9) we have

$$E(Y(\infty)|T_{s} = 1, X) =$$

$$Pr(T_{u} = 1|T_{s} > 1, X)E(Y|T_{s} > 1, T_{u} = 1, X) +$$

$$Pr(T_{u} > 1|T_{s} > 1, X)E(Y|T_{s} > 2, T_{u} > 1, X).$$
(10)

That is, the long-run outcomes for non-treated units with $T_u = 1$ and $T_u > 1$ are weighted in a specific way, based on the exit probabilities $Pr(T_u = 1 | T_s > 1, X)$, and this allows us to recover the long-run outcome under no treatment. This deals with the fact that some of the non-treated become treated in the second period.

4.2 Identification results

The first part of the derivation below concisely generalizes the line of reasoning in the previous subsection for an arbitrary number of periods. We show that $E(Y(\infty)|T_s = t_s, T_u(t_s) \ge t_s)$ (i.e., the second component of ATET (t_s)) is identified under our assumptions. Note that the first component, $E(Y(t_s)|T_s = t_s, T_u(t_s))$, is given by (4).

We first rewrite the counterfactual outcome using Assumptions 1 and 2,

$$E(Y(\infty) \mid T_s = t_s, T_u(t_s) \ge t_s) = E(Y(\infty) \mid T_s = t_s, T_u(\infty) \ge t_s).$$

$$(11)$$

Next, we condition on X and apply Assumption 3 (assignment) for period t_s :

$$E(Y(\infty) \mid T_s = t_s, T_u(\infty) \ge t_s, X) = E(Y(\infty) \mid T_s > t_s, T_u(\infty) \ge t_s, X).$$
(12)

Then, by the law of iterated expectations

$$E(Y(\infty) \mid T_s > t_s, T_u(\infty) \ge t_s, X) =$$

$$Pr(T_u = t_s \mid T_s > t_s, T_u(\infty) \ge t_s, X) E(Y(\infty) \mid T_s > t_s, T_u(\infty) = t_s, X) +$$

$$Pr(T_u > t_s \mid T_s > t_s, T_u(\infty) \ge t_s, X) E(Y(\infty) \mid T_s > t_s, T_u(\infty) > t_s, X).$$
(13)

That is, the counterfactual outcome under "never treatment" can be decomposed into average outcomes for individuals with $T_u = t_s$ and for individuals with $T_u > t_s$. The former, with $T_u = t_s$, in (13) are non-treated individuals who leave the initial state directly in period t_s . For this group, the observational rule in (1) and Assumptions 1 and 2 give

$$E(Y(\infty)|T_s > t_s, T_u(\infty) = t_s, X) = E(Y|T_s > t_s, T_u = t_s, X).$$
(14)

Also, note that the treatment probabilities $\Pr(T_u = t_s | T_s > t_s, T_u(\infty) \ge t_s, X) = \Pr(T_u = t_s | T_s > t_s, T_u \ge t_s, X)$ and $\Pr(T_u > t_s | T_s > t_s, T_u(\infty) \ge t_s, X) = \Pr(T_u > t_s | T_s > t_s, T_u \ge t_s, X)$ are observed.

For the other group, with $T_u > t_s$, in (13), i.e., for those who survive at least one

additional time period, we have

$$\begin{split} E(Y(\infty)|T_{s} > t_{s}, T_{u}(\infty) > t_{s}, X) &= E(Y(\infty)|T_{s} > t_{s} + 1, T_{u}(\infty) > t_{s}, X) = \\ E(Y(\infty)|T_{s} > t_{s} + 1, T_{u}(\infty) \ge t_{s} + 1, X), \end{split}$$

where the first equality follows from Assumption 3 for period $t_s + 1$, and the second equality from re-writing. Next, using (13) by replacing t_s with $t_s + 1$ we have

$$E(Y(\infty) \mid T_s > t_s + 1, T_u(\infty) \ge t_s + 1, X) =$$
(15)

$$\begin{aligned} &\Pr(T_u = t_s + 1 | T_s > t_s + 1, T_u(\infty) \ge t_s + 1, X) E(Y(\infty) | T_s > t_s + 1, T_u(\infty) = t_s + 1, X) + \\ &\Pr(T_u > t_s + 1 | T_s > t_s + 1, T_u(\infty) \ge t_s + 1, X) E(Y(\infty) | T_s > t_s + 1, T_u(\infty) > t_s + 1, X). \end{aligned}$$

For the sake of presentation, introduce some auxiliary notation for the distribution of T_u :

$$h(t,X) = \Pr(T_u = t \mid T_s > t, T_u \ge t, X).$$

Using this notation, (13)-(15) and (14) for period $t_s + 1$ give

$$\begin{split} E(Y(\infty)|T_s > t_s, T_u(\infty) \geq t_s, X) &= h(t_s, X) & E(Y|T_s > t_s, T_u = t_s, X) + \\ & [1 - h(t_s, X)]h(t_s + 1, X) & E(Y|T_s > t_s + 1, T_u = t_s + 1, X) + \\ & [1 - h(t_s, X)][1 - h(t_s + 1, X)] & E(Y(\infty)|T_s > t_s + 1, T_u(\infty) > t_s + 1, X). \end{split}$$

All parts of this expression are observed except $E(Y(\infty)|T_s > t_s + 1, T_u(\infty) > t_s + 1, X)$. However, for this outcome, we can iteratively use (13) and (14) for $t_s + 2, ..., T_u^{\text{max}}$, where T_u^{max} is maximum time in the initial state. This gives:

$$E(Y(\infty)|T_s > t_s, T_u(\infty) \ge t_s, X) = \sum_{k=t_s}^{T_u^{\text{max}}} h(k, X) \prod_{m=t_s}^{k-1} [1 - h(m, X)] E(Y|T_s > k, T_u = k, X).$$
(16)

Interestingly, this is the expectation of the random outcome variable Y over a discretetime competing risks duration distribution. In this interpretation, the competing risks are treatment and exit out of the initial state, and only the observed outcomes after the second risk (exit) are used.

Next, note that

$$E(Y(\infty)|T_s = t_s, T_u(t_s) \ge t_s) = E_{X|T_s = t_s, T_u \ge t_s} [E(Y(\infty)|T_s = t_s, T_u(t_s) \ge t_s, X)].$$
(17)

Then, $E(Y(\infty) | T_s = t_s, T_u(t_s) \ge t_s)$ is identified from (11)-(17). This gives the second component of ATET(t_s), and (4) gives the first component of ATET(t_s). This is summarized in Theorem 1.

Theorem 1 (ATET) If Assumptions 1–3 hold then

 $ATET(t_s) = E(Y|T_s = t_s, T_u \ge t_s) -$

$$E_{X|T_s=t_s,T_u\geq t_s}\left[\sum_{k=t_s}^{T_u^{\max}} h(k,X) \prod_{m=t_s}^{k-1} [1-h(m,X)] E(Y|T_s>k,T_u=k,X)\right],$$

where

$$h(t,X) = \Pr(T_u = t | T_s > t, T_u \ge t, X).$$

In sum, the non-treated group (consisting of individuals leaving the initial state before becoming treated or actually never would have become treated) can be used to identify the counterfactual outcome for those treated after a certain elapsed duration. Theorem 1 shows that this is achieved by giving individuals leaving the initial state a differential weight depending on the duration in that state.

The above derivation demonstrates the independent roles of the unconfoundedness assumption and the no-anticipation assumption. The unconfoundedness assumption re-

lates to the allocation of treatment across individuals and assures that the treated and the not-yet treated have similar potential outcomes. The no-anticipation assumption concerns the relationship between different potential outcomes for a given individual, and assures that the outcomes for the not-yet treated can be used to mimic the outcomes under never treatment even if some of the non-yet treated may be treated if they remain in the initial state.

Identification of $ATE(t_s)$ follows using similar reasoning as above. For the second component of $ATE(t_s)$, our assumptions give

$$E(Y(\infty)|T_u(t_s) \ge t_s) = E_{X|T_u \ge t_s} \left[E(Y(\infty)|T_u(\infty) \ge t_s, X) \right] =$$

$$E_{X|T_u \ge t_s} \left[E(Y(\infty)|T_s > t_s, T_u(\infty) \ge t_s, X) \right],$$
(18)

and from (16):

$$E(Y(\infty)|T_s > t_s, T_u(\infty) \ge t_s, X) = \sum_{k=t_s}^{T_{us}^{\max}} h(k, X) \prod_{m=t_s}^{k-1} [1 - h(m, X)] E(Y|T_s > k, T_u = k, X).$$
(19)

Thus, from (18)-(19) we have

$$E(Y(\infty)|T_u(t_s) \ge t_s) =$$
⁽²⁰⁾

$$E_{X|T_{u} \ge t_{s}} \left[\sum_{k=t_{s}}^{T_{u}^{\max}} h(k,X) \prod_{m=t_{s}}^{k-1} [1-h(m,X)] E(Y|T_{s} > k, T_{u} = k,X) \right].$$

For the first component of $ATE(t_s)$:

$$E(Y(t_s)|T_u(t_s) \geq t_s) = E(Y(t_s)|T_u(\infty) \geq t_s)$$

$$= E_{X|T_u \geq t_s} [E(Y(t_s)|T_u(\infty) \geq t_s, X)]$$

$$= E_{X|T_u \geq t_s} [E(Y(t_s)|T_s = t_s, T_u(\infty) \geq t_s, X)]$$

$$= E_{X|T_u \geq t_s} [E(Y|T_s = t_s, T_u \geq t_s, X)],$$
(21)

where the first equality follows from Assumptions 1 and 2, the second by the law of iterated expectations, the third by Assumption 3 and the fourth equality from the observational rule and Assumptions 1 and 2.

From (20) and (21) we thus obtain,

Theorem 2 (ATE) If Assumptions 1–3 hold then

$$ATE(t_s) = E_{X|T_u \ge t_s} \left[E(Y|T_s = t_s, T_u \ge t_s, X) \right] -$$

$$E_{X|T_{u} \ge t_{s}} \left[\sum_{k=t_{s}}^{T_{u}^{\max}} h(k,X) \prod_{m=t_{s}}^{k-1} [1-h(m,X)] E(Y|T_{s} > k, T_{u} = k,X) \right]$$

where

$$h(t,X) = \Pr(T_u = t | T_s > t, T_u \ge t, X).$$

 $ATET(t_s)$ and $ATE(t_s)$ can be aggregated over the distribution of T_s to obtain the overall average effect of the program. Specifically,

$$ATET = E_{t_s}[E(Y(t_s) - Y(\infty) \mid T_s = t_s, T_u(t_s) \ge t_s)]$$

where the outer expectation is taken over the distribution of T_s . A similar expression applies to ATE. In this context it is important to point out that all quantities above refer to (and will be quantified by data from) a world in which the program exists. In particular, before individuals are treated, they may be aware of the existence of the program and act accordingly. Assumption 1 stipulates that such ex ante behavior is equal for all individuals until the treatment occurs. Still, the behavior in the various control groups used for inference on counterfactual outcomes may be affected by such ex ante behavior. For example, the existence of a training program may reduce efforts to find jobs before treatment occurs, for every unemployed individual in the market. As such, the over-all effects ATET and ATE capture effects relative to "no treatment" in a world where the program exists.

5 Estimation

We rely on the constructive identification results of the previous section to construct estimators of the average treatment effects of interest. Specifically, in Appendix A it is shown that if Assumptions 1–3 hold, then an unbiased estimator of $ATET(t_s)$ is provided by:

$$\widehat{\text{ATET}}(t_s) = \frac{1}{\rho_{t_s} N_{t_s}} \sum_{i \in T_{s,i} = t_s, T_{u,i} \ge t_s} Y_i -$$
(22)

$$\frac{1}{\sum\limits_{i\in T_{s,i}>T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)}\sum_{i\in T_{s,i}>T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)Y_i,$$

where N_t is the number of non-treated survivors at the beginning of t and

$$w^{t_{s}}(t_{u},X) = \frac{p(t_{s},X)}{\rho_{t_{s}}[1-p(t_{s},X)]} \frac{1}{\prod_{m=t_{s+1}}^{t_{u}}[1-p(m,X)]}$$
(23)
$$p(t,X) = \Pr(T_{s}=t|T_{s} \ge t, T_{u} \ge t,X)$$

$$\rho_{t} = \Pr(T_{s}=t|T_{s} \ge t, T_{u} \ge t),$$

in practice the weights, *w*, are replaced by estimated weights, \hat{w} , with estimated treatment probabilities (propensity scores), \hat{p} . For ATE(*t_s*) we have by the results in Appendix A

$$\widehat{\text{ATE}}(t_{s}) = \frac{1}{\sum_{i \in T_{s,i} = t_{s}, T_{u,i} \ge t_{s}} w_{ATE1}^{t_{s}}(X_{i})} \sum_{i \in T_{s,i} = t_{s}, T_{u,i} \ge t_{s}} w_{ATE1}^{t_{s}}(X_{i})Y_{i} -$$

$$\frac{1}{\sum_{i \in T_{s,i} > T_{u,i} \ge t_{s}} w_{ATE0}^{t_{s}}(T_{u,i},X_{i})} \sum_{i \in T_{s,i} > T_{u,i} \ge t_{s}} w_{ATE0}^{t_{s}}(T_{u,i},X_{i})Y_{i},$$
(24)

where

$$w_{ATE1}^{t}(X) = \frac{1}{p(t,X)}$$

$$w_{ATE1}^{t_s}(t_u,X) = \frac{1}{1-p(t_s,X)} \frac{1}{\prod_{m=t_{s+1}}^{t_u} [1-p(m,X)]}$$

This follows the ideas of Horvitz and Thomson (1952) and weights the outcome responses of the treated and non-treated toward the target population (either the treated population at t_s or the full population of survivors at t_s). Both (22) and (24) are based on normalized weights. The reason for this is that without the normalization the weights not always add up to one.

Consider the intuition behind the estimator $ATE(t_s)$ in (24). Only non-treated individuals, provide information about the counterfactual outcome under never treatment for those treated at t_s . However, for the reasons discussed above, the outcomes of the nontreated need to be weighted for several reasons. The first part of the weights, $\frac{1}{1-p(t_s,X_i)}$, follows from IPW estimators in the static evaluation literature (see e.g., Wooldridge, 2010). Under Assumption 3 this adjusts for covariate differences between the treated t_s and those still waiting for treatment at t_s .

The second part of the weights corrects for the fact that some non-treated at t_s start treatment at $t_s + 1, ...$ As a background, consider the case with only treatment assignment at t_s , so that everyone who do not receive treatment at t_s remains untreated in all time periods. We focus on non-normalized weights. Then, in large samples:

$$\widehat{\text{ATE}}(t_s) = \frac{1}{N_{t_s}} \sum_{i \in T_{s,i} = t_s, T_{u,i} \ge t_s} \frac{Y_i}{p(t_s, X_i)} - \frac{1}{N_{t_s}} \sum_{i \in T_{s,i} > T_{u,i} \ge t_s} \frac{Y_i}{1 - p(t_s, X_i)} = \frac{1}{N_{t_s}} \sum_{i \in T_{s,i} \ge t_s, T_{u,i} \ge t_s} \left[\frac{I[T_s = t_s, T_u \ge t_s]Y_i}{p(t_s, X_i)} - \frac{I[T_s > t_s, T_u \ge t_s]Y_i}{1 - p(t_s, X_i)} \right],$$

i.e. the same structure and weights as static IPW estimators. The average effect is obtained by re-weighting the responses of the treated and not treated at t_s . Notice that for the second part of this expression, we have conditional on *X*

$$E\left[\frac{I[T_{s} > t_{s}, T_{u} \ge t_{s}]Y}{[1 - p(t_{s}, X)]} | X\right] = E\left[\frac{I[T_{s} > t_{s}, T_{u} \ge t_{s}]Y(\infty)}{1 - p(t_{s}, X)} | X\right] = \frac{[1 - p(t_{s}, X)]E[Y(\infty)|X, T_{s} > t_{s}, T_{u} \ge t_{s}]}{1 - p(t_{s}, X)} = E[Y(\infty)|T_{s} > t_{s}, T_{u} \ge t_{s}, X].$$

Next,

$$E[Y(\infty)|T_s > t_s, T_u \ge t_s, X] =$$

$$\sum_{t=t_s}^{T_u^{\max}} \Pr(T_u = t|T_s > t_s, T_u \ge t_s) E[Y(\infty)|T_s > t_s, T_u = t, X],$$
(25)

i.e. the average outcomes of the untreated survivors at t_s could be expressed as an average over the average outcome of the non-treated for each duration between t_s and T_u^{max} . Thus, with only treatment assignment at t_s the outcomes of the not treated are weighted using the IPW weights from the static evaluation problem. Following (25), this is an average that gives individuals with the same X the same weight regardless of time in the initial state. With treatment assignment at t_s, \ldots additional weighting is necessary, because in this case, individuals with a high treatment probability and/or individuals remaining in the initial state for a long time are less likely to remain non-treated. The IPW estimators above correct for this, since the weights depend on both the observed characteristics and the time in the initial state.

To estimate the over-all average effect ATET aggregated over all possible t_s , we average over the distribution of T_s , where the fraction treated after t is equal to $N_t / \sum_{m=1}^{T_u^{\text{max}}} N_m$. Specifically, we average $\widehat{\text{ATET}}(t_s)$ over the effects for specific pre-treatment durations. This gives:

$$\widehat{\text{ATET}} = \sum_{t=1}^{T_u^{\text{max}}} \frac{N_t}{\sum_{m=1}^{T_u^{\text{max}}} N_m} \frac{1}{\rho_t N_t} \sum_{i \in T_{s,i} = t, T_{u,i} \ge t} Y_i, -$$
(26)
$$\sum_{t=1}^{T_u^{\text{max}}} \frac{N_t}{\sum_{m=1}^{T_u^{\text{max}}} N_m} \frac{1}{\sum_{i \in T_{s,i} > T_{u,i} \ge t} w^t (T_{u,i}, X_i)} \sum_{i \in T_{s,i} > T_{u,i} \ge t} w^t (T_{u,i}, X_i) Y_i.$$

A similar approach can be used to estimate ATE.

It is well known that IPW estimation may be sensitive to extreme values of the propensity scores, since single observations may receive too large weight (see e.g., Frölich 2004, Huber et al. 2013, Busso et al. 2014). One way to overcome this problem is trimming. One trimming approach is the three-step approach in Huber et al. (2013). For average treatment effects, their approach imposes zero weights to all treated (controls) whose share of the sum of all original weights in the treatment (control) group is larger than c% (e.g., 4%). Thereafter, normalize the weights again and finally discard observations whose propensity score is smaller (greater) than the maximum (minimum) of the minimum (maximum) scores among the treated and controls. This assures that extreme values are discarded and that the same type of individuals are discarded in both treatment arms. This approach is also applicable to our estimator, noting that in our case the weights are a function of several propensity scores.

Concerning inference, (22) and (24) are similar to the IPW estimator proposed by Hirano et al. (2003) and also discussed by Lunceford and Davidian (2004). One important difference is that here, the weights depend on several propensity scores instead of a single propensity score. Nevertheless, if the propensity scores are known, the large-sample variances for $\widehat{ATE}(t_s)$ can be derived using similar reasoning as in Lunceford and Davidian (2004). More generally, Hirano et al. (2003) show that the variance decreases if estimated scores are used instead of true scores. In that case, standard errors can be obtained by bootstrapping.

A simulation study on the properties of the estimator is given in Appendix B. The simulations show that the bias of our new estimator is virtually zero in all simulations, the bootstrap estimator for the standard errors has correct size, and the standard standard error decreases by roughly 50% when the sample size is increased by a factor of four (estimator is \sqrt{N} -convergent). We also compare with a static IPW estimator, which as expected, is biased, and the bias is increasing in, for instance, the share of treated.

6 Extensions

6.1 Short-run and long-run outcomes

So far we have considered effects on the long-run outcome realized after the individuals left the initial state. We now consider effects on both short-run and long-run outcomes. Specifically, define Y_t as the observed outcome in period t. The corresponding potential outcomes are denoted by $Y_t(t_s)$. Here, the object of interest is the average treatment effect of treatment at t_s on the outcome in period $t_s + \tau$ (i.e., τ periods after the start of the treatment) among the treated at t_s :

$$ATET(t_s, \tau) = E(Y_{t_s+\tau}(t_s) - Y_{t_s+\tau}(\infty) | T_s = t_s, T_u(t_s) \ge t_s).$$

$$(27)$$

For training programs for unemployed workers, this may be the effect on earnings a certain number of periods after the start of the training, so that $ATET(t_s, \tau)$ for smaller values of τ capture any lock-in effects during the program, and subsequent $ATET(t_s, \tau)$ reflect the effects in the medium-run and the long-run.

In Section 2 we introduced the no-anticipation Assumptions 1 and 2. Since Y is now a short-term outcome, we now need a no-anticipation assumption for Y_t that is similar to Assumption 1 for T_u ,

Assumption 4 (No anticipation, short-run outcomes)

$$E[Y_t(t')] = E[Y_t(t'')], t < \min(t',t'').$$

For completeness we also generalize the unconfoundedness assumption:

Assumption 5 (Assignment, short-run outcomes) For all t,

$$P_t \perp \{Y_t(t_s), \ldots, Y_{\infty}(t_s)\}_t^{\infty} \mid X, T_s \geq t, T_u \geq t.$$

The observation rule is now

$$Y_{t} = \sum_{m=1}^{t} [\mathbf{I}(T_{s} = m)Y_{t}(m)] + \mathbf{I}(T_{s} > \min(T_{u}, t))Y_{t}(\infty),$$
(28)

which holds in the absence of right-censoring of T_u and by invoking Assumptions 1 and 4. The first part of the right-hand side of (28) states that if the individual is treated before t, then we observe the potential outcome corresponding to the actually observed time to

treatment.⁶ The second part states that if an individual have left the initial state before t without having been treated or if the individual is treated after t, then the observed outcome at t is the outcome corresponding to the assignment "never treated".

The first component of $ATET(t_s, \tau)$ is identified from the observed outcomes, Y_t , of those treated at time t_s :

$$E(Y_{t_s+\tau}(t_s)|T_s=t_s, T_u(t_s) \ge t_s) = E(Y_{t_s+\tau}|T_s=t_s, T_u \ge t_s).$$
⁽²⁹⁾

For the second component of $ATET(t_s, \tau)$, condition on *X*. Then, by Assumptions 1, 4 and 5,

$$E(Y_{t_s+\tau}(\infty)|T_s=t_s, T_u(t_s) \ge t_s, X) = E(Y_{t_s+\tau}(\infty)|T_s=t_s, T_u(\infty) \ge t_s, X)$$
(30)
$$= E(Y_{t_s+\tau}(\infty)|T_s>t_s, T_u(\infty) \ge t_s, X),$$

and by the law of iterated expectations and the observational rule

$$E(Y_{t_{s}+\tau}(\infty)|T_{s}>t_{s},T_{u}(\infty)\geq t_{s},X) = h(t_{s},X)E(Y_{t_{s}+\tau}(\infty)|T_{s}>t_{s},T_{u}(\infty)=t_{s},X) + (31)$$
$$[1-h(t_{s},X)]E(Y_{t_{s}+\tau}(\infty)|T_{s}>t_{s},T_{u}(\infty)>t_{s},X),$$

where $E(Y_{t_s+\tau}(\infty)|T_s > t_s, T_u(\infty) = t_s, X) = E(Y_{t_s+\tau}|T_s > t_s, T_u = t_s, X)$, and the probability $h(t_s, X)$ also is observed. By Assumption 5 for period $t_s + 1$ and (31) by replacing t_s with $t_s + 1$ we have

$$E(Y_{t_s+\tau}(\infty)|T_s > t_s, T_u \ge t_s, X) = E(Y_{t_s+\tau}(\infty)|T_s > t_s+1, T_u(\infty) \ge t_s+1, X) =$$
(32)

$$h(t_s+1,X)E(Y_{t_s+\tau}(\infty)|T_s>t_s+1,T_u(\infty)=t_s+1,X)$$

⁶In applications, if the time unit in the data is large, then it is possible that the observed Y_t (say, total earnings in t) for $t = T_s$ reflects to some extent the earnings within period t that were obtained in the sub-period prior to the treatment, and, indeed, that a realization of T_s in the middle of period t depends on earnings earlier in that same time period. This is a commonly encountered challenge in applied discrete-time dynamic treatment analyses. In our setting it can be dealt with by allowing the conditioning set X to be time-varying (see the next subsection for this approach) but only if the data provide outcomes on a finer grid than the time unit in the data. Of course it depends on the institutional setting as well as on the data whether the treatment assignment can give rise to such concerns.

+
$$[1-h(t_s+1,X)]E(Y_{t_s+\tau}(\infty)|T_s>t_s+1,T_u(\infty)>t_s+1,X),$$

where the first equality follows from Assumption 5 and the second from (31). Iteratively, for $t_s + 2, ..., t$ gives

$$E(Y_{t_s+\tau}(\infty)|T_s > t_s, T_u(\infty) \ge t_s, X) =$$

$$\sum_{k=t_s}^{t_s+\tau} h(k,X) \prod_{m=t_s}^{k-1} [1 - h(m,X)] E(Y|T_s > k, T_u = k, X) +$$

$$\prod_{m=t_s}^{t_s+\tau} [1 - h(m,X)] E(Y_{t_s+\tau}|T_s > t_s + \tau, T_u > t_s + \tau, X).$$
(33)

Combining (30) and (33) and averaging over X gives the second component of (27). All these results are in Theorem 3.

Theorem 3 (ATET with short-run and long-run outcomes) *If Assumptions 1, 4 and 5 hold then*

$$\begin{aligned} \text{ATET}(t_s, \tau) &= E(Y_{t_s + \tau} | T_s = t_s, T_u \ge t_s) - \\ E_{X|T_s = t_s, T_u \ge t_s} [\sum_{k=t_s}^{t_s + \tau} h(k, X) \prod_{m=t_s}^{k-1} [1 - h(m, X)] E(Y|T_s > k, T_u = k, X) - \\ &\prod_{m=t_s}^{t_s + \tau} [1 - h(m, X)] E(Y_{t_s + \tau} | T_s > t_s + \tau, T_u > t_s + \tau, X)], \end{aligned}$$

where

$$h(t,X) = \Pr(T_u = t | T_s > t, T_u \ge t, X).$$

We can also extend the estimation results in a similar way. In Appendix A, we show that if Assumptions 1, 4 and 5 hold an unbiased estimator of $ATET(t_s)$ is:

$$\widehat{\text{ATET}}(t_s) = \frac{1}{\rho_{t_s} N_{t_s}} \sum_{i \in T_{s,i} = t_s, T_{u,i} \ge t_s} Y_{t_s + \tau, i} -$$
(34)

$$\frac{1}{\rho_{t_s}N_{t_s}}\left[\sum_{i\in T_{s,i}>T_{u,i},t_s+\tau\geq T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)Y_{t_s+\tau,i}+\sum_{i\in T_{s,i}>t_s+\tau,T_{u,i}>t_s+\tau}w^{t_s}_{\tau}(T_{u,i},X_i)Y_{t_s+\tau,i}\right],$$

where $w^{t_s}(t_u, X)$ is given by (23) and

$$w_{\tau}^{t_s}(X) = \frac{p(t_s, X)}{\prod_{m=t_s}^{t_s+\tau} 1 - p(m, X)}.$$

6.2 Time-varying covariates

We now allow for selection on *time-varying* observed covariates. In evaluations of active labor market programs, this allows treatment assignments to not only depend on characteristics measured at the beginning of the unemployment spell, but also on characteristics that change during the unemployment spell. For instance, some unemployed workers divorce and local labor market conditions can change during the unemployment spell.

To rule out effects of the treatment in period t on X, the covariates determining treatment assignment in t should be measured before assignments are made. For that reason we use the notation X_{t^-} for the observed covariates at t, where t^- indicates that X is measured at least slightly before t. Note that X_{t^-} may include covariates from previous periods, even the entire vector of covariates from all previous periods. By analogy to Assumption 1, we require a "no-anticipation" assumption regarding future changes in the time-varying covariates.

With time-varying covariates our sequential unconfoundedness assumption is:

Assumption 6 (Assignment time-varying covariates) For all t,

$$[P_t \perp \{Y(t_s)\}_t^{\infty} \mid X_{t^-}, T_s \ge t, T_u(\infty) \ge t]$$

As above, the identification problems concerns the counterfactual outcome $E(Y(\infty)|T_s = t_s, T_u(t_s) \ge t_s)$. Here, similar reasoning as above can be used. The main difference is that we now first average over $X_{t_s}^-$, then over $X_{t_s+1}^-$ and so on. Specifically, we have the following result:⁷

⁷The standard overlap condition for treatment assignment now is that for all *t* and $X_{t^-} \Pr(P_t = 1 | T_s \ge t, T_u \ge t, X_{t^-}) < 1$, taking censoring into account.

Theorem 4 (ATET with time-varying covariates) If Assumptions 1, 2 and 6 hold then

$$\begin{aligned} \text{ATET}(t_s) &= E(Y|T_s = t_s, T_u \ge t_s) - \\ & [E_{X_{t_s}^-|T_s = t_s, T_u \ge t_s}[h(X_{t_s}^-)E(Y|T_s > t_s, T_u = t_s, X_{t_s}^-) + \\ & [1 - h(X_{t_s}^-)]E_{X_{t_s+1}^-|T_s > t_s, T_u > t_s, X_{t_s}^-}[h(X_{t_s+1}^-)E(Y|T_s > t_s + 1, T_u = t_s + 1, X_{t_s+1}^-) + \\ & [1 - h(X_{t_{s+1}}^-)]E_{X_{t_s+2}^-|T_s > t_s + 1, T_u > t_s + 1, X_{t_s+1}^-}[h(X_{t_{s+2}}^-)E(Y|T_s > t_s + 2, T_u = t_s + 2, X_{t_s+2}^-) + \dots + \end{aligned}$$

$$[1 - h(X_{T_u^{\max}-1}^{-})]E_{X_{T_u^{\max}}^{-}|T_s > T_u^{\max}-1, T_u > T_u^{\max}-1, X_{T_u^{\max}-1}^{-}}[h(X_{T_u^{\max}}^{-}) \times E(Y|T_s > T_u^{\max}, T_u = T_u^{\max}, X_{T_u^{\max}}^{-})]\dots]]]].$$

where

$$h(X_t^-) = \Pr(T_u = t | T_s > t, T_u \ge t, X_t^-).$$

Proof See Appendix A.

These identification results show that average effects are identified with time-varying covariates. In terms of estimation, Appendix A shows that if Assumptions 1 and 2 and 6 hold an unbiased estimator of $ATET(t_s)$ is:

$$\widehat{\text{ATET}}(t_s) = \frac{1}{\rho_{t_s} N_{t_s}} \sum_{i \in T_{s,i} = t_s, T_{u,i} \ge t_s} Y_i - \frac{1}{\rho_{t_s} N_{t_s}} \sum_{i \in T_{s,i} > T_{u,i} \ge t_s} w^{t_s}(T_{u,i}, X_i^-) Y_i$$
(35)

where

$$w^{t_s}(t_u, X^-) = \frac{p(t_s, X_{t_s}^-)}{\prod_{m=t_s}^{t_u} [1 - p(m, X_m^-)]}$$

6.3 Right-censored durations

We now allow for right-censoring of the durations in the eligibility state. It is important to explain what we mean with this in the context of a long-run post-spell outcome. First, the right-censoring can only take place while the unit is in the eligibility state. Let T_c denote the (possibly individual-specific) time until the unit is right-censored. At each discrete

elapsed duration in the eligibility state, the binary event governing whether $T_c = t$ vs. $T_c > t$ is realized before the binary events governing whether $T_s = t$ vs. $T_s > t$ and $T_u = t$ vs. $T_u > t$. This implies that at a given time unit, censoring is realized before treatment assignment. If $T_c = t$, any treatment realizations and exits at t and after t are unobserved. Note that we can write $T_c = \infty$ if $T_c > T_u$.

Secondly, the long-run outcome is assumed to be unobserved if the unit is rightcensored, i.e., if $T_c \leq T_u$. This introduces another selection problem because any rightcensored unit does not carry any information about the long-run outcomes and the durations of certain types of individuals may be censored at a higher rate. Since both treated and non-treated durations can be censored, this problem occurs for both non-treated units and the counterfactual potential outcome under no treatment, as well as for the treated units and the treated potential outcomes.

The above type of right-censoring is common in many applications. For instance, individuals may drop out from the study, or there may be a limited follow-up period after the initial sample is drawn. In the case of unemployment durations, individuals may drop out due to emigration, mortality, or exit out of the labor force. Individuals who do not enter employment after a certain time period may be transferred to welfare payments and thus end up in different administrative registers.

Since it is assumed that the realization of the right-censoring outcome is the first event within each period, we now have the following average treatment effect of t_s on Y among those who are actually treated at t_s :

$$ATET(t_s) = E(Y(t_s) - Y(\infty) \mid T_s = t_s, T_c > t_s, T_u(t_s) \ge t_s).$$
(36)

We adjust for right-censoring in a similar way as for treatment assignments and consider identification under unconfounded right-censoring. Formally, let the binary indicator $C_t := 1$ iff $T_c = t$. The assumption is:

Assumption 7 (Right-censored process) For all t,

$$\left[C_t \bot \{Y(t_s)\}_t^{\infty} \mid X, T_s \ge t, T_c \ge t, T_u \ge t\right],$$

where, as before, the joint distribution of $\{Y(t_s)\}_t^{\infty}$ is taken to cover all values $t_s = t, t + 1, ..., \infty$. This assumption echoes the unconfoundedness assumption for treatment assignments and rules out that there are systematic unobserved characteristics that drive both the right censoring and the long-run outcome. As for Assumption 3, the plausibility of Assumption 7 depends on the exact censoring process and on the information available.⁸

Using similar reasoning as above, we arrive at Theorem 5,

Theorem 5 (ATET with right-censored durations) If Assumptions 1–3 and 7 hold then

 $ATET(t_s) =$

$$E_{X|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s}}\left[\sum_{k=t_{s}}^{T_{u}^{\max}}h_{c1}(t,X,t_{s})\prod_{m=t_{s}}^{k-1}[1-h_{c1}(t,X,t_{s})]E(Y|T_{s}=t_{s},T_{c}>k,T_{u}=k,X)\right]-E_{X|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s}}\left[\sum_{k=t_{s}}^{T_{u}^{\max}}h_{c}(k,X)\prod_{m=t_{s}}^{k-1}[1-h_{c}(m,X)]E(Y|T_{s}>k,T_{c}>k,T_{u}=k,X)\right].$$

where

$$h_{c1}(t, X, t_s) = \Pr(T_u = t \mid T_s = t_s, T_c > t, T_u \ge t, X).$$

$$h_c(t, X) = \Pr(T_u = t \mid T_s > t, T_c > t, T_u \ge t, X).$$

Proof See Appendix A.

These identification results show that the average effects are identified with rightcensoring. The only difference is that the right-censoring introduces another selection problem, so that in addition to selection due treatment assignments, there is also selection due to right-censored durations.

In terms of estimation, in Appendix A it is shown that if Assumptions 1–3, and 7 hold,

⁸An overlap condition for the censoring process is also needed. For all t and X $\Pr(C_t = 1 | T_c \ge t, T_u \ge t, X) < 1$

^{1.} Formally, the standard overlap condition for treatment assignment now is: for all t and X $Pr(P_t = 1 | T_s \ge t, T_c > t, T_u \ge t, X) < 1$, taking censoring into account.

then an unbiased estimator of $ATET(t_s)$ is:

$$\widehat{\text{ATET}}(t_s) = \frac{1}{\sum_{i \in T_{s,i} = t_s, T_{c,i} > T_{u,i}, T_{u,i} \ge t_s} w_{c_1}^{t_s}(T_{u,i}, X_i)} \sum_{i \in T_{s,i} = t_s, T_{c,i} > T_{u,i}, T_{u,i} \ge t_s} w_{c_1}^{t_s}(T_{u,i}, X_i) Y_i - (37)$$

$$\frac{1}{\sum_{i \in T_{s,i} > T_{u,i}, T_{c,i} > T_{u,i}, T_{u,i} \ge t_s} w_{c_0}^{t_s}(T_{u,i}, X_i)} \sum_{i \in T_{s,i} > T_{u,i}, T_{c,i} > T_{u,i}, T_{u,i} \ge t_s} w_{c_0}^{t_s}(T_{u,i}, X_i) Y_i,$$

where $N_{t_s}^c$ is the number of non-treated survivors at the beginning of t_s with durations censored after t_s and

$$\begin{split} w_{c_1}^{t_s}(t_u, X) &= \frac{1}{\prod_{m=t_{s+1}}^{t_u} [1 - e_{c1}(m, t_s, X)]} \\ w_{c_0}^{t_s}(t_u, X) &= \frac{p_c(t_s, X)}{\rho_{t_s}^c [1 - p_c(t_s, X)] \prod_{m=t_s+1}^{t_u} [1 - p_c(m, X)] [1 - e_{c0}(m, X)]} \\ \rho_t^c &= \Pr(T_s = t | T_u \ge t, T_c > t, T_s \ge t) \\ p_c(t, X) &= \Pr(T_s = t | T_s \ge t, T_c > t, T_u \ge t, X) \\ e_{c1}(t_c, t_s, X) &= \Pr(T_c = t_c | T_s = t_s, T_c \ge t_c, T_u \ge t_c, X) \\ e_{c0}(t, X) &= \Pr(T_c = t | T_s \ge t, T_c \ge t, T_u \ge t, X). \end{split}$$

7 WIA training in the US

7.1 Background

The Workforce Investment Act (WIA) of 1998 replaced the Job Training Partnership Act (JTPA). The WIA program provides federal resources for employment services at the state and the local level. One important goal of WIA was to consolidate services for unemployed workers. The coordinated services now take place at around 2,500 One-Stop centers (American Job Centers) nationwide. All participation in WIA is voluntary, but caseworkers at the local job center need to authorize all staff-assisted services, including training. The services are provided through three funding streams aimed at adult workers, dislocated workers, and youths.⁹

⁹Adults are individuals 18 years satisfying certain criteria, dislocated workers are comprise adults who recently lost a job, and the youth program target youths in ages 18-21.

The services are offered at three levels. All WIA participants receive core services, which include staff-assisted job search, placement and labor market information, and basic counseling. The second level involves intensive services with more comprehensive assessment and counseling, career planning, and in some case shorter courses.¹⁰ The third level involves training services, such as on-the-job training, apprenticeships, training at vocational schools, and community colleges. The idea is that the participants should start with the core services and if no job is found proceed to the intensive services and/or training, However, in a national evaluation of the implementation of WIA, D'Amico et al. (2004) find extensive local and state variation in the implementation of WIA. Some participants follow the intended sequence with more and more intensified services, while others almost immediately proceed to training services.¹¹

Several previous studies evaluate the service offered within the WIA program. In a recent study, Andersson et al. (2016) compare WIA participants who receive training to other WIA participants, and estimate the effect of training conditional on WIA participation. Their preferred estimates (data for 1999–2005) reveal significantly positive effects of WIA training on earnings and employment for workers served under the adult funding stream, but mostly negative effects for dislocated workers. Heinrich et al. (2013) find positive effects on earnings from WIA training for workers in the WIA adult program. However, for dislocated workers they find negative short-run effects and modest positive effects in the long-run. Hollenbeck et al. (2005) conclude that WIA services increase employment rates and that WIA training generates positive effects in comparison with other WIA services. Ao et al. (2016) study the effects of the length of WIA participation. Finally, McConnell et al. (2016) report evidence from a randomized trial with. One result from the 15-month impact evaluation is that WIA training did not increase earnings when compared to other WIA services.

¹⁰The implementation of the intensive services vary across local area. Some areas provide a rich set of intensive services, while other areas mainly use the intensive services as preparation for training and/or exclusively as individualized job search assistance.

¹¹D'Amico et al. (2004) and Barnow and King (2005) provide detailed descriptions of WIA.

7.2 The dynamic evaluation setting

We follow Andersson et al. (2016) and Heinrich et al. (2013) and study the impact of WIA training conditional on WIA registration. That is, we compare WIA participants who receive training and perhaps also other types of services to WIA participants who received only core and/or intensive services. In this setting, there are two durations. First, T_u , is time in the WIA program, i.e. WIA registration time. An individual becomes eligible for WIA services when (s)he enters WIA and remains eligible while in the WIA program. In a sense this resembles the classic European training program eligibility where individuals typically are eligible for training while unemployed. Second, the treatment duration, T_s , is the time from WIA registration to training. The outcome of interest, Y, is earnings measured some time after the WIA training.

Note that the WIA selection process is a dynamic process. Some participants start with core services and then proceed with training, while other participants start training in the first or second month after WIA registration. This gives rise to extensive variation in the time from WIA registration to training, which is illustrated in *Figure 1*, which shows the distribution of the time to WIA training in months.¹² The figure shows that a rather large fraction of all training episodes start in the first month after WIA registration, but 23.1% of the training participants wait more than three months before they enroll in training. Those who start training after several months are workers who remain on WIA services after several months, most likely, are a selected sample of less successful participants. The implication is that treatment status as either treated (WIA training) or non-treated (WIA services but no training) depends on the time in WIA, which creates the dynamic treatment assignment problem addressed in this paper.

7.3 Data and estimation details

To evaluate WIA training, the WIASRD public use files are used. It contains information on all WIA participants. We select all WIA registration spells that started between April 1, 2011 and March 31, 2013. We follow previous studies and focus on the adult program and the dislocated workers program, and consider the effect of the first training program

¹²When WIA was initially implemented it placed large focus on a first-work policy, but later guidelines placed more emphasizes on training (Barnow and King, 2005). This is why training often starts in the first months.



Figure 1: Distribution of time to WIA training, by months since WIA registration

within the first 12 months in WIA. As outcomes we use the available information in the WIASRD public use files and study wages over four quarters after WIA.

The WIA selection process involves both WIA participants and the WIA staff. WIA participants' desires for training will depend on, for instance, their age, motivation, ability to learn, perception of the job market and beliefs about the value of training. All these factors will also influence the actions taken by the WIA staff. They will also take the budget set of the local WIA into account. Based on detailed information on this selection process and using insights from studies, such as Lechner and Wunsch (2013) and Caliendo et al. (2017), that compare alternative sets of conditioning variables, Andersson et al. (2016) select a number of variables to control for in the analysis. This includes variables that are directly observed (e.g., age), variables that are proxies for motivation and ability (e.g., preprogram wages), and geographic indicators to control for the state of the local economy. All this allows Andersson et al. (2013) to argue that unconfoundedness holds.¹³ They also argue that other WIA participants represent a natural comparison group for WIA training participants, and that there is a substantial amount of WIA caseworker discretion over the different WIA services, which lends further support to the unconfoundedness assumption.

¹³Andersson et al. (2016) also compare their main selection on observed variables with difference-indifferences type of estimators that control for time invariant unobserved heterogeneity, but find very small differences.
Heinrich et al. (2013) use a similar set of background characteristics and also evaluate under unconfoundedness.

We also analyze the WIA training based on unconfounded treatment assignment. The conditioning variables include sex, race (3 categories), age (10 categories), education (4 categories), disability indicator, veteran indicator, three calendar quarters of pre-WIA wages, registration year and quarter indicators, and a full set of state dummies. Compared to Andersson et al. (2016), the main difference is that we do not have information on geographic location within each state (the local One-Stop office at which the participant enrolled), because this information is excluded from the public use files. This is unfortunate, since Andersson et al. (2016) finds that controlling for the geographic location of the WIA offices is important. Since our analysis is restricted to WIASARD administrative data we also have somewhat less information on pre-WIA wages.^{14,15}

Table 1 presents sample statistics for training participants and the untreated. The sample includes roughly 1.5 million WIA registration spells of which 231,540 (13.0%) concern participation in training. The reported background characteristics show that there are more females, blacks and younger workers among the training participants.

Concerning estimation details, our dynamic IPW estimator involves estimation of a series of propensity scores, one for each time period. These propensity scores are estimated using logistic regression models. Due to the lower number of treated after four months on WIA, the propensity scores are estimated jointly from the fourth month and onwards (with indicators for each duration month). Standard errors are obtained using bootstrap (99 replications).¹⁶

We compare the dynamic estimator to a static IPW approach. Here, treated individuals

¹⁴Andersson et al. (2016) also use matched employer data on from the Longitudinal Employer-Household Dynamics to obtain characteristics of the last employer before registration, including information on industry, size and turnover rates. They conclude that the additional firm variables do not pass a researcher cost-benefit test in this context based on what they add in terms of the credibility of the CIA.

¹⁵Besides unconfoundedness, no-anticipation also needs to hold. Since there are several sources of uncertainty in the assignment process we believe that this assumption holds. For instance, as already mentioned, WIA staff have large influence over enrollment decisions, making it difficult for the workers to predict the exact timing of the training.

¹⁶We also explore common support restrictions using a variant of the three-step approach in Huber et al. (2013). We set an upper bound on the weight given to the outcome for a certain individual (1%), but due to the rather large sample size, this common support restriction is not very binding.

	No training	Training
# observations	1551313	231540
Mean survival time	4.9	12.4
Outcomes		
Earnings first quarter after WIA	3626	5355
Earnings second quarter after WIA	3864	5622
Earnings third quarter after WIA	3910	5698
Earnings fourth quarter after WIA	3888	5778
Background characteristics		
Male	54.0	48.0
Race		
White	62.1	53.4
Black	19.5	26.4
Other	0.2	0.2
Age at registration		
-20	5.6	4.9
20–25	14.4	15.4
26–30	14.0	14.6
31–35	12.2	13.0
36–40	10.5	11.5
41–45	10.7	11.5
46–50	11.0	11.2
51–55	10.2	9.8
56–60	7.6	6.2
Education		
Less than High School	3.1	0.9
High School	58.6	53.2
Some College	24.1	32.7
College or more	14.2	13.2
Disabled	4.1	2.5
Veteran	7.6	7.0
Pre-program earnings, quarter before registration		
First	4815.5	3837.0
Second	5029.3	4407.9
Third	5026.8	4682.9
Year of registration		
2011	34.7	36.7
2012	53.6	52.3
2013	11.8	11.0
Quarter of registration		
First	28.4	25.3
Second	23.0	26.0
Third	24.5	26.3
Fourth	24.1	22.3

 Table 1: Sample statistics for WIA training participants and other WIA participants

Note: Background characteristics recorded at registration. Earnings in 2012 US dollars.

are those who obtain training within the first 12 months in WIA. We adjust for differences between treated and non-treated using a standard IPW approach:

$$\delta_{StaticIPW} = \frac{1}{n_1} \sum_{i=1}^{N} W_i Y_i - \frac{1}{n_0} \sum_{i=1}^{N} \left(\frac{1}{n_0} \sum_{i=1}^{N} \frac{\hat{e}(x_i)(1-W_i)}{1-\hat{e}(x_i)} \right)^{-1} \frac{\hat{e}(x_i)(1-W_i)Y_i}{1-\hat{e}(x_i)}, \quad (38)$$

where $\hat{e}(X_i)$ is the estimated scores for the probability that W = 1, n_1 the number of treated observations and n_0 the number of untreated observations.

7.4 Estimation results

Table 2 presents the main estimation results. The first column provides a raw comparison of the treated and the non-treated workers without adjusting for any observed variables. It shows that training participants on average have \$1737 higher earnings than if they had been non-treated, for the first quarter after WIA. Adjusting for observed variables using a static IPW estimator reduces the estimate to \$1008 (Column 2).

	Raw [1]	Static IPW [2]	Dynamic IPW [3]	Difference [3]-[2] [4]	Diff. in percent [5]	p-value (Diff.) [6]
First quarter	1737	1008	1233	225	22.3	0.000
	(13.3)	(18.8)	(25.5)	(18.2)		
Second quarter	1767	1138	1353	215	18.9	0.000
	(13.2)	(18.0)	(24.2)	(18.3)		
Third quarter	1794	1176	1393	217	18.4	0.000
	(15.5)	(21.3)	(24.4)	(17.0)		
Fourth quarter	1897	1219	1397	178	14.6	0.000
	(15.3)	(24.4)	(28.5)	(19.6)		

Table 2: Effects of WIA training on earnings

Note: Quarterly earnings in 2012 US dollars. Dynamic IPW is the dynamic IPW estimator introduced in this paper. Raw makes no covariate adjustments and Static IPW is a standard static IPW estimator with normalized weights.

With our dynamic IPW estimator, this estimate increases to \$1233, which is 22.3% larger than the static IPW estimates, and this difference is significant (Columns 3-6).¹⁷ All this confirms the expected patterns. With a static approach, the control group includes

¹⁷The standard error for the difference between the two estimates is obtained using bootstrap.

too many participants who only remain in WIA for a short period. This is, most likely, a group with more favorable characteristics, and this leads to an effect estimate that is biased downwards. Our dynamic approach corrects for this positive selection of the control group.

Table 2 also reveals similar patterns for earnings in the second, third, and fourth quarters after exit from WIA.

8 Long-run effects of Swedish training programs

8.1 Background and data

The main aim of the Swedish training program called AMU is to improve the skills of unemployed workers. The training courses in this program are directed towards the upgrading of skills or the acquisition of skills that are in short supply or that are expected to be in short supply. The most common courses offer training towards manufacturing (11.6% of the participants), machine operators (9.8%), office/warehouse work (15.1%), health care (6.1%) and computer skills (15.1%).¹⁸

While the training program is for unemployed workers, it is important to know the long-run effects on earnings. This fits into the dynamic evaluation problem studied in this paper. For treated workers, the long-run earnings are unobserved, and under the unconfoundedness assumption the treated and the not-yet treated at *t* are comparable. However, some non-treated receive treatment in subsequent periods, and this creates a dynamic evaluation problem because some workers have a higher treatment probability, and since for a given period-by-period individuals with a high rate of leaving unemployment are more likely to remain non-treated.

Several register-based datasets are used in the analysis. From the Swedish employment offices, we have day-by-day information on the unemployment status and on participation in training and other labor markets programs. These data also include a number of personal characteristics. We use gender, age dummies (3 categories), level of education (3 categories), regional of residence (6 areas), inflow year dummies, and indicators

¹⁸Evaluations of the effect of the program on unemployment durations include Harkman and Johansson (1999) and van den Berg and Richardson (2013).

for UI entitlement and if the unemployed only search for a job in the local area. The employment office data is also used to construct information on unemployment history. Additional background characteristics are obtained from an annual population register, including indicators for children in different age brackets, marital status, and labor earnings history. The population register also includes information on long-run labor earnings (all cash compensation paid by employers, consumer price adjusted) up to ten years after the training.

For estimation reasons, the daily data is aggregated into monthly intervals. To be able to study long-run effects, we sample all unemployment spells that start in the period 1995–1998. We focus on unemployed individuals with no previous unemployment record within 180 days before the unemployment spell, and restrict the analysis to workers in ages 25–55 at the time of entry into unemployment.¹⁹ We consider the effect of the first training program during the unemployment. All propensity scores are estimated using logistic regression models and standard errors are obtained by bootstrapping (99 replications).²⁰

Table 3 presents sample statistics. In total, the analysis sample includes 735,581 unemployment spells of which 47,411 (6.1%) concern participation in the training program. We also see that there are more females, high-school educated, university educated, married, and parents among the participants.

As in the WIA application, the analysis rests on an unconfoundedness assumption. We control for socio-economic characteristics, previous income and unemployment, and regional indicators. This selection of variables is based on several previous studies. Heckman et al. (1998, 1999) stress that it is important to control for previous unemployment, lagged earnings, and local labor market conditions. More recently, Lechner and Wunsch (2013) and Biewen et al. (2014) obtain similar results. Here, we adjust for both previous unemployment and earnings, and regional labor market conditions (through regional dummies). As further support of the unconfoundedness assumption, Eriksson (1997) and

¹⁹The reason for the latter is that the benefits entitlement rules and active labor market policy programs were different for persons aged below 25 or above 55 during the period studied here. Individuals below 25 must participate in a program after 100 days of unemployment, or otherwise they lose their unemployment benefits. They can use special programs that are not available for other age groups. Persons over 55 receive unemployment benefits for 450 days.

²⁰When using the three-step common support approach as in Section 7, we obtain similar results.

	Untreated	Treated	
# observations	735581	47411	
Mean survival time	8.7	23.8	
Outcomes			
Earnings program year	69354	56865	
Earnings $+1$ years	79886	50721	
Earnings +2 years	101121	84456	
Earnings $+3$ years	119704	106003	
Earnings +4 years	133278	120001	
Earnings +5 years	142343	129455	
Earnings +6 years	147861	135164	
Earnings +7 years	152404	138984	
Earnings +8 years	158090	143800	
Background characteristics			
Male	47 A	47 8	
Age at the start of the spell	77.7	41.0	
25_34	57.8	50.9	
35-44	26.4	30.5	
45 <u>-</u> 54	15.8	18.6	
Married	36.5	40.7	
Eduction	00.0	10.1	
Less than High school	0.2	0.2	
High school education	51.9	58.0	
University education	26.4	21 7	
Child in ages 0-3	25.6	30.1	
Child in ages 4-6	17.7	19.2	
Child in ages 7-15	24.7	27.6	
Ul eligiable	78.0	80.6	
Rest, search area	16.6	19.4	
Pre-program earnings and unemployment	2010	2000	
Days unemployed year -1	58.0	63.8	
Days unemployed year -2	81.1	94.1	
Earnings vear -1	72092	73124	
Earnings year -2	74197	73128	
Earnings year -3	77117	76035	
Year of inflow			
1996	24.9	20.7	
1997	24.6	19.1	
1998	26.0	26.6	
Area of residence			
Stockholm MSA	21.1	17.4	
Gothenburg MSA	16.4	13.7	
Skane MSA	13.3	12.9	
North	14.2	15.3	
South	11.8	13.4	

Table 3: Sample statistics for Swedish labor market training participants and non-participants

Note: Controls recorded at the start of the unemployment spell. Earnings is in SEK.

Carling and Richardson (2001) show that caseworkers have a large influence and large degree of discretionary power over enrollment decisions, which suggests that individual self-selection into the training program is less of a problem.

8.2 Estimation results

The main results are presented in *Table 4*. Column 3 reports the results from our dynamic estimator. It shows that in earnings, on average, decrease by about 14,500 SEK in the year that the program starts. This is because of so called lock-in effects during the program (Gerfin and Lechner, 2002, van Ours, 2004, and Sianesi, 2004). In the long-run, the effects are much more positive with sizable and persistent positive effects on yearly labor earnings. For instance, five years after the program, the effect is 6543 SEK or 4.6%.

	Raw [1]	Static IPW [2]	Dynamic IPW [3]	Difference [3]-[2] [4]	Diff. in percent [5]	p-value (Diff.) [6]
Prog. year	-27823	-27756	-14455	13301	47.9	0.000
0,	(266.0)	(281.2)	(280.8)	(410.9)		
Prog.+1 years	-14279	-12282	-5196	7086	57.7	0.000
	(365.4)	(376.4)	(461.5)	(519.5)		
Prog. $+ 2$ years	-7525	-4405	7295	11700	265.6	0.000
	(442.7)	(436.4)	(613.8)	(576.3)		
Prog. $+ 3$ years	-7799	-3999	8899	12899	322.5	0.000
	(483.3)	(453.1)	(720.8)	(673.9)		
Prog. $+ 4$ years	-9019	-5084	7697	12782	251.4	0.000
	(497.0)	(475.2)	(794.1)	(683.1)		
Prog. $+$ 5 years	-9274	-5699	6543	12242	214.8	0.000
	(534.7)	(508.2)	(754.1)	(671.9)		
Prog. $+ 6$ years	-10127	-6616	5223	11839	178.9	0.000
	(578.8)	(557.2)	(746.2)	(745.9)		
Prog. $+7$ years	-10976	-7162	5000	12162	169.8	0.000
	(567.8)	(568.7)	(785.1)	(875.5)		
Prog. $+$ 8 years	-11531	-7255	5357	12612	173.8	0.000
	(619.1)	(594.4)	(888.8)	(1002.8)		
Prog. $+$ 9 years	-11805	-6904	6005	12909	187.0	0.000
	(608.5)	(577.4)	(1019.8)	(1165.3)		

Table 4: Effects of Swedish AMU training on yearly labor earnings

Note: Outcome labor earnings in SEK. Dynamic IPW is the dynamic IPW estimator introduced in this paper. Raw makes no covariate adjustments and Static IPW is a standard static IPW estimator with normalized weights. Bootstrapped standard errors (99 replications)

Table 4 also reveals striking differences between a static IPW approach (Column 2)

and our dynamic IPW approach. For instance, five years after the training, the dynamic approach gives a positive training effect, while the static approach suggests a large negative effect of training. Needles to say, this has very different policy implications. It also means that adjusting for the dynamic treatment assignment is relatively more important for this Swedish training program than for the WIA training program, although the differences is sizeable and significant also in the WIA case. There are two explanations to this. First, *Figure 2* shows that a larger share of the participants in the Swedish training program start training after many months of unemployment. Since the static approach do not adjust for differences in pre-treatment durations, the control group includes too many short-term unemployed workers. This leads to substantial downward bias as short-term unemployed workers tend to have more favourable characteristics. For WIA training, a larger share starts training in the first months (see *Figure 1*), and this leads to a somewhat less pronounced dynamic evaluation problem.





Second, in the Swedish case, the abundance of short-term unemployed workers in the static approach's control group is especially problematic because of the large observed earnings differences between short-term and long-term unemployed workers. Average earnings five years after the unemployment spell are about 185,000 for the previously short-term unemployed (less than four months of unemployment) and 129,000 for the

previously long-term unemployed (more than 12 months of unemployment). In the WIA case, individuals who participate in WIA for only a short period of time are more similar to individuals who stay a bit longer, which makes for a less pronounced dynamic problem.

9 Conclusions

This paper presents new identification results and proposes new estimators for treatment evaluations based on unconfoundedness assumptions in a dynamic setting. We apply this to programs from both the US and Sweden. The US application evaluates the WIA program and compares participation in WIA training with participation in other WIA services. Here, the initial state is being registered in the WIA program and the dynamic evaluation problem arises because the date of entry into the training varies. In the Swedish application, we estimate the long-run earnings effects of a flagship training program for unemployed workers in Sweden. Training programs for unemployed workers is a classic dynamic evaluation setting, where training occurs after different elapsed unemployment durations.

In both applications, we find positive earnings effects of training, and taking the dynamic treatment assignment into account turns out to be important. For the WIA program, a standard static evaluation approach underestimates the WIA training effect by roughly 20% when it is compared to our new dynamic approach. For the Swedish program, the difference between a static and our dynamic approach is even more striking as the static approach leads to estimates with an incorrect sign (negative effect instead of positive effects on earnings in the long-run). Indeed, for reasonable discount rates, our results lead to a reassessment of the net over-all benefits of the program.

Until recently, the literature as well as policy debates seem to have agreed on a consensus that training programs are ineffective (see surveys by e.g. Heckman, Smith and LaLonde, 1999, and Crépon and Van den Berg, 2016). However, much of the evidence was based on short-run observed outcomes and/or methodologies that could not appropriately deal with the dynamic assignment of training. Our empirical analyses, based on novel data and a new method, suggest that this consensus may have to be overturned. At the very least, our results cautiously suggest that there is scope for a reappraisal of training.

The analyses in this paper do not explicitly distinguish between the direct causal effect of the treatment and an indirect effect that may run through the length of stay in the initial eligibility state. Often one would expect the former to dominate in the longer run. However, we view it as an interesting topic for further research to develop a formal statistical framework for mediation analysis that distinguishes between the two channels.

References

- Abbring J.H. and G.J. van den Berg (2003), "The non-parametric identification of treatment effects in duration models", *Econometrica*, 71, 1491–1517.
- Ao W., S. Calonico and Y.Y. Lee (2016), "Evaluating the Effects of Lengths of Participation in the Workforce Investment Act Adult Program via Decomposition Analysis", mimeo, Institute for Defense Analyses.
- Andersson F, H.J Holzer, J.I Lane, D Rosenblum and J. Smith (2016), "Does Federally-Funded Job Training Work? Nonexperimental Estimates of WIA Training Impacts Using Longitudinal Data on Workers and Firms", CESifo Working Paper Series 6071, CESifo Group Munich.
- Barnow B.S. and C.T. King (2005), "The Workforce Investment Act in Eight States", Report Prepared for U.S. Department of Labor, Employment and Training Administration.
- Biewen M., B. Fitzenberger, A. Osikominu and M. Paul (2014), "The Effectiveness of Public Sponsored Training Revisited: The Importance of Data and Methodological Choices", *Journal of Labor Economics* 32(4), 837—897.
- Busso M., J. DiNardo and J. McCrary (2014), ""New Evidence on the Finite Sample Properties of Propensity Score Reweighting and Matching Estimators", *Review of Economics and Statistics* 96(5), 885—897.
- Caliendo M., R. Mahlstedt and O. Mitnik (2017), "Unobservable, but unimportant? The relevance of usually unobserved variables for the evaluation of labor market policies", *Labour Economics* 46, 14–25.
- Carling K. and K. Richardson, K. (2001), "The Relative Efficiency of Labor Market Programs: Swedish Experience From the 1990", *Labour Economics*, 26:4, 335–354.
- Crépon, B., M. Ferracci, G. Jolivet and G.J. van den Berg (2009), "Active Labor Market Policy Effects in a Dynamic Setting", *Journal of the European Economic Association*,

7, 595–605.

- Crépon, B. and G.J. van den Berg (2016), "Active Labor Market Policies Setting", Annual Review of Economics 8(1), 521–546.
- D'Amico, R., K. Dunham, J. Henderson-Frakes, D. Kogan, V. Koller, M. Mack, M. Magnotta, J. Salzman et al. (2004), "The Workforce Investment Act after Five Years: Results from the National Evaluation of the Implementation of WIA. Oakland", California: Social Policy Research Associates.
- Eriksson, M. (1997), "To choose or not to choose: Choice and choice set models", Umeå Economic Studies 443, Department of Economics, UmeåUniversity.
- Fitzenberger B., A. Osikominu and R. Völter (2008), "Get Training or Wait? Long-Run Employment Effects of Training Programs for theUnemployed in West Germany", *Annales d'Économie et de Statistique*, 91/92, 321-355.
- Forslund A. and J. Vikström (2011) "Arbetsmarknadspolitikens effekter på sysselsättning och arbetslöshet en översikt (The effects of labor market policies on employment and unemployment)", IFAU Report 2011:7.
- Frölich M. (2004), "Finite Sample Properties of Propensity-Score Matching and Weighting Estimators", *Review of Economics and Statistics* 86, 77–90.
- Gerfin M. and M. Lechner (2002), "A Microeconometric Evaluation of the Active Labour Market Policy in Switzerland", *Economic Journal*, 112, 854–893.
- Ham, J.C. and R.J. LaLonde (1996), "The effect of sample selection and initial conditions in duration models: Evidence from experimental data on training", *Econometrica*, 64, 175–205.
- Harkman A. and Johansson A. (1999), "Training or subsidized jobs—what works?", Working paper, AMS, Solna.
- Heckman J.J, H. Ichimura and P. Todd (1998), "Matching As An Econometric Evaluation Estimator", *Review of Economic Studies* 65, 261–294.

- Heckman, J., Lalonde, R., and Smith, J. (1999), "The Economics and Econometrics of Active Labor Market Programs", in *Handbook of Labor Economics*, Vol. 3, eds. O. Ashenfelter and D. Card, Amsterdam: North-Holland, pp. 1865–2097.
- Heinrich C.J, P.R. Mueser, K.R. Troske, K.S. Jeon and D.C. Kahvecioglu (2013), "Do Public Employment and Training Programs Work?", *IZA Journal of Labor Economics*, 2:6
- Hirano K., G. Imbens and G. Ridder (2003), "Efficient estimation of average treatment effects using the estimated propensity score", *Econometrica*, 71, 1161–1189.
- Hollenbeck K., D. Schroeder, C.T. King and W. Huang (2005), "Net impact estimates for services provided through the Workforce Investment Act". In: On the use of administrative data for workforce development program evaluation, ETA Occasional Paper 2005–06. U.S. Department of Labor, Employment and Training Administration.
- Horovitz D.G. and D.J. Thompson (1952), "A Generalization of Sampling Without Replacement From a Finite Universe", *Journal of the American Statistical Association*, 47, 663–685.
- Huber M., M. Lechner, and C. Wunsch (2013), "The Performance of Estimators Based on the Propensity Score", *Journal of Econometrics* 175, 1–21.
- Imbens and Wooldridge (2009), "A Generalization of Sampling Without Replacement From a Finite Universe", *Journal of the American Statistical Association*, 47, 663– 685.
- Lechner M. (1999), "Earnings and employment effects of continuous off-the-job training in East Germany after unification", *Journal of Business and Economic Statistics*, 17, 74-90.
- Lechner M. (2002), "Some practical issues in the evaluation of heterogenous labour market programmes by matching methods", *Journal of Royal Statistical Society A* 165(1), 59-82.

- Lechner M. and B. Melly (2010), "Partial Identification of Wage Effects of Training Programs", Working Papers 2010-8, Brown University, Department of Economics.
- Lechner M., R. Miquel and C. Wunsch (2011), "Long-run Effects of Public Sector Sponsored Training in West Germany", *Journal of European Economic Association*, 9(4), 742–784.
- Lechner M. and C. Wunsch (2013), "Sensitivity of matching-based program evaluations to the availability of control variables," *Labour Economics* 21, 111–121.
- Lee D.S. (2009), "Training, Wages, and Sample Selection: Estimating Sharp Bounds on Treatment Effects", *Review of Economic Studies*, 76, 1071–1102.
- Lunceford J.K. and Davidian M. (2004), "Stratification and weighting via the propensity score in estimation of causal treatment effects: a comparative study", *Statistics in Medicine*, 23, 2937–2960.
- McConnell S., K. Fortson, D. Rotz, P. Schochet, P. Burkander, L. Rosenberg and A. Mastri (2016), "Providing Public Workforce Service to Job Seekers: 15-Month Impact Findings on the WIA Adult and Dislocated Worker Programs", Mathematica Policy Research Report, May 2016.
- Muller P., B. van der Klaauw and A. Heyma (2017), "Comparing Econometric Methods to Empirically Evaluate Job-Search Assistance", IZA Discussion Papers, No. 10531.
- PES (2016), "Annual labor market report 2016 (Arbetsmarknadsrapport 2016)", Working paper, Swedish Public Employment Office (Arbetsförmedlingen), Stockholm (in Swedish).
- Regnér, H. (2002), "A nonexperimental evaluation of training programs for the unemployed in Sweden", *Labour Economics*, 9, 187–206.
- Richardson K. and G.J. van den Berg (2013), "Duration dependence versus unobserved heterogeneity in treatment effects: Swedish labor market training and the transition rate to employment", *Journal of Applied Econometrics*, 28, 325–351.

- Sianesi B. (2004), "An Evaluation of the Swedish System of Active Labour Market Programmes in the 1990s", *Review of Economics and Statistics*, 86, 133–155.
- Sianesi B. (2008), "Differential effects of active labour market programs for the unemployed", *Labour Economics*, 15(3), 370–399.
- Vikström J. (2017), "IPW estimation and related estimators for evaluation of active labor market policies in a dynamic setting", *Labour Economics* 49, 42–54.
- Wooldridge J.M. (2010). *Econometric Analysis of Cross Section and Panel Data (2nd ed.)*, MIT Press.
- Wunsch C. and M. Lechner (2008), "What Did All the Money Do? On the General Ineffectiveness of Recent West German Labour Market Programmes", *Kyklos*, 61(1), 134–174.

Appendix A. Proofs and derivations

Estimation of ATET(*t*_s)

We show that if Assumptions 1–3 hold, the IPW estimator, $\widehat{\text{ATET}}$, is an unbiased estimator of $\text{ATET}(t_s) = E(Y(t_s) - Y(\infty) \mid T_s = t_s, T_u(t_s) \ge t_s)$.

For the first part of $ATET(t_s)$, the estimator without the normalization is:

$$\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}=t_s,T_{u,i}\geq t_s}Y_i,$$

for which we have

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}=t_s,T_{u,i}\geq t_s}Y_i\right] =$$

$$\frac{1}{\rho_{t_s}}E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}\geq t_s,T_{u,i}\geq t_s}\mathbf{I}(T_{s,i}=t_s)Y_i\right] =$$

$$\frac{1}{\rho_{t_s}}E\left[\mathbf{I}(T_s=t_s)Y|T_s\geq t_s,T_u\geq t_s\right] =$$

$$\frac{1}{\rho_{t_s}}\Pr(T_s=t_s|T_u\geq t_s,T_s\geq t_s)E\left[Y|T_s=t_s,T_u\geq t_s\right] =$$

$$E\left[Y|T_s=t_s,T_u\geq t_s\right] =$$

$$E\left[Y(t_s)|T_s=t_s,T_u(t_s)\geq t_s\right],$$
(A.1)

where the last equality follows by Assumptions 1 and 2 and the observational rule. Note that $\rho_{t_s} = \Pr(T_s = t_s | T_s \ge t_s, T_u \ge t_s)$.

For the second part of $ATET(t_s)$, the estimator without the normalization is:

$$\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)Y_i,\tag{A.2}$$

for which we have

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)Y_i\right] = \\E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}\geq t_s,T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)\mathbf{I}(T_{s,i}>T_{u,i})Y_i\right] =$$

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}\geq t_s, T_{u,i}\geq t_s}\sum_{t_u=t_s}^{T_u^{\max}}w^{t_s}(t_u, X_i)\mathbf{I}(T_{s,i} > t_u, T_{u,i} = t_u)Y_i\right] = \frac{1}{\rho_{t_s}}E\left[\sum_{t_u=t_s}^{T_u^{\max}}w^{t_s}(t_u, X)\mathbf{I}(T_s > t_u, T_u = t_u)Y|T_s \ge t_s, T_u \ge t_s\right] = E_{X|T_s\geq t_s, T_u\geq t_s}\left[\frac{1}{\rho_{t_s}}E\left[\sum_{t_u=t_s}^{T_u^{\max}}w^{t_s}(t_u, X)\mathbf{I}(T_s > t_u, T_u = t_u)Y|T_s \ge t_s, T_u \ge t_s, X\right]\right].$$

For sake of presentation, use the notation

$$h(t,X) = \Pr(T_u = t | T_s > t, T_u \ge t, X).$$

Next, using Assumptions 1–3, and that $w^{t_s}(t_u, X) = \frac{p(t_s, X)}{\prod_{m=t_s}^{t_u} [1-p(m, X)]}$:

$$E\left[w^{t_s}(t_u, X)\mathbf{I}(T_s > t_u, T_u = t_u)Y|T_s \ge t_s, T_u \ge t_s, X\right] =$$
(A.3)

$$w^{t_{s}}(t_{u},X) \operatorname{Pr}(T_{s} > t_{u}, T_{u} = t_{u} | T_{s} \ge t_{s}, T_{u} \ge t_{s}, X) E[Y|T_{s} > t_{u}, T_{u} = t_{u}, X] =$$

$$\frac{p(t_{s},X)}{\prod_{m=t_{s}}^{t_{u}} [1 - p(m,X)]} h(t_{u},X) \prod_{m=t_{s}}^{t_{u}-1} [1 - h(m,X)] \prod_{m=t_{s}}^{t_{u}} [1 - p(m,X)] E[Y|T_{s} > t_{u}, T_{u} = t_{u}, X] =$$

$$p(t_{s},X) h(t_{u},X) \prod_{m=t_{s}}^{t_{u}-1} [1 - h(m,X)] E[Y|T_{s} > t_{u}, T_{u} = t_{u}, X] =$$

$$p(t_{s},X) h(t_{u},X) \prod_{m=t_{s}}^{t_{u}-1} [1 - h(m,X)] E[Y(\infty)|T_{s} > t_{u}, T_{u}(\infty) = t_{u}, X] =$$

$$p(t_{s},X) h(t_{u},X) \prod_{m=t_{s}}^{t_{u}-1} [1 - h(m,X)] E[Y(\infty)|T_{s} = t_{s}, T_{u}(\infty) = t_{u}, X] =$$

Note that the second equality follows from the definition of $w^{t_s}(t_u, X)$, the third equality by simplifying, the fourth equality by Assumptions 1 and 2, and the fifth equality by applying Assumption 3 for $t_s, ..., t_u$.

From (A.2) and (A.3)

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i},T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)Y_i\right] =$$
(A.4)

$$E_{X|T_{s}\geq t_{s},T_{u}\geq t_{s}}\left[\frac{p(t_{s},X)}{\rho_{t_{s}}}\sum_{t_{u}=t_{s}}^{T_{u}^{\max}}h(t_{u},X)\prod_{m=t_{s}}^{t_{u}-1}[1-h(m,X)]E[Y(\infty)|T_{s}=t_{s},T_{u}(\infty)=t_{u},X]\right].$$

For sake of presentation, introduce the notation

$$y(T_u(\infty) = t, X) = E[Y(\infty)|T_s = t_s, T_u(\infty) = t, X]$$
$$y(T_u(\infty) > t, X) = E[Y(\infty)|T_s = t_s, T_u(\infty) > t, X]$$
$$y(T_u(\infty) \ge t, X) = E[Y(\infty)|T_s = t_s, T_u(\infty) \ge t, X].$$

Using this notation we have using that by construction $h(T_u^{\max}, X) = 1$

$$h(T_u^{\max}, X) \prod_{m=t_s}^{T_u^{\max}-1} [1 - h(m, X)] y(T_u(\infty) = T_u^M, X) = \prod_{m=t_s}^{T_u^{\max}-1} [1 - h(m)] y(T_u(\infty) = T_u^M, X).$$
(A.5)

Next, for time periods $T_u^{\text{max}} - 1$ and $T_u^{\text{max}} - 2$

$$\prod_{m=t_{s}}^{T_{u}^{\max}-1} [1-h(m,X)]y(T_{u}(\infty) = T_{u}^{M},X) + h(T_{u}^{M}-1,X)] \prod_{m=t_{s}}^{T_{u}^{\max}-2} [1-h(m,X)]y(T_{u}(\infty) = T_{u}^{M}-1,X) =$$
(A.6)
$$\prod_{m=t_{s}}^{T_{u}^{\max}-2} [1-h(m,X)]y(T_{u}(\infty) \ge T_{u}^{M}-1,X),$$

and for arbitrary time periods t and t - 1

$$\prod_{m=t_s}^{t} [1-h(m,X)]y(T_u(\infty) > t,X) + h(t)] \prod_{m=t_s}^{t-1} [1-h(m,X)]y(T_u(\infty) = t-1,X) =$$
(A.7)
$$\prod_{m=t_s}^{t-1} [1-h(m,X)]y(T_u(\infty) \ge t-1,X).$$

Thus, using (A.5) for T_u^{max} , (A.6) for $T_u^{\text{max}} - 1$ and (A.7) for $t_s, ..., T_u^{\text{max}} - 2$ we have

$$\sum_{t_u=t_s}^{T_u^{\max}} h(t_u, X) \prod_{m=t_s}^{t_u-1} [1 - h(m, X)] E[Y(\infty)|T_s = t_s, T_u(\infty) = t_u, X] =$$
(A.8)
$$\sum_{t_u=t_s}^{T_u^{\max}} h(t_u, X) \prod_{m=t_s}^{t_u-1} [1 - h(m, X)] y(T_u(\infty) = t_u, X) \stackrel{(A.5)}{=}$$

$$\begin{split} \prod_{m=t_s}^{T_u^{\max}-1} [1-h(m,X)] y(T_u(\infty) &= T_u^{\max}, X) + \sum_{t_u=t_s}^{T_u^{\max}-1} h(t_u,X) \prod_{m=t_s}^{t_u-1} [1-h(m,X)] y(T_u(\infty) = t_u,X) \stackrel{(A.6)}{=} \\ \prod_{m=t_s}^{T_u^{\max}-2} [1-h(m,X)] y(T_u(\infty) &\geq T_u^{\max}-1, X) + \sum_{t_u=t_s}^{T_u^{\max}-2} h(t_u,X) \prod_{m=t_s}^{t_u-1} [1-h(m,X)] y(T_u(\infty) = t_u,X) \stackrel{(A.7)}{=} \\ y(T_u(\infty) &\geq t_s, X). \end{split}$$

Thus, from (A.4) and (A.8)

$$E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}, T_{u,i}\geq t_s}w^{t_s,ATE}(T_{u,i},X_i)Y_i\right] =$$

$$E_{X|T_s\geq t_s, T_u\geq t_s}\left[\frac{p(t_s,X)}{\rho_{t_s}}y(T_u(\infty)\geq t_s,X)\right] =$$

$$\frac{1}{\rho_{t_s}}E_{X|T_s\geq t_s, T_u\geq t_s}\left[p(t_s,X)E[Y(\infty)|T_s=t_s,T_u(\infty)\geq t_s,X]\right] =$$

$$\frac{1}{\rho_{t_s}}E_{X|T_s\geq t_s, T_u\geq t_s}\left[\Pr(T_s=t_s|T_s\geq t_s,T_u\geq t_s,X)E[Y(\infty)|T_s=t_s,T_u(\infty)\geq t_s,X]\right] =$$

$$\frac{1}{\rho_{t_s}}\Pr(T_s=t_s|T_s\geq t_s,T_u\geq t_s)E[Y(\infty)|T_s=t_s,T_u(\infty)\geq t_s] =$$

$$E[Y(\infty)|T_s=t_s,T_u(\infty)\geq t_s].$$
(A.9)

This averaging over X is admitted by the common support assumption.

Finally, (A.1) and (A.9) imply that $E\left[\widehat{\text{ATET}}(t_s)\right] = \text{ATET}(t_s)$.

Estimation of ATE(t_s)

We now show that if Assumptions 1–3 hold, $\widehat{ATE}(t_s)$ is an unbiased estimator of $ATE(t_s) = E(Y(t_s) - Y(\infty) | T_s \ge t_s, T_u(\infty) \ge t_s).$

For the first part of $ATE(t_s)$, the estimator without the normalization is:

$$\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}=t_s,T_{u,i}\geq t_s}w_{ATE1}^{t_s}(X_i)Y_i,$$

under Assumptions 1-3 we have

$$E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}=t_s, T_{u,i}\geq t_s}w_{ATE1}^{t_s}(X_i)Y_i\right] =$$
(A.10)

$$E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}\geq t_s, T_{u,i}\geq t_s}w_{ATE1}^{t_s}(X_i)\mathbf{I}(T_{s,i}=t_s)Y_i\right] =$$
$$E\left[w_{ATE1}^{t_s}(X)\mathbf{I}(T_s=t_s)Y|T_s\geq t_s, T_u\geq t_s\right] =$$
$$E_{X|T_s\geq t_s, T_u\geq t_s} \left[E\left[w_{ATE1}^{t_s}(X)\mathbf{I}(T_s=t_s)Y|T_s\geq t_s, T_u\geq t_s, X\right]\right] =$$
$$E_{X|T_s\geq t_s, T_u\geq t_s} \left[\frac{1}{p(t_s, X)}p(t_s, X)E\left[Y|X, T_s=t_s, T_u\geq t_s, X\right]\right] =$$
$$E_{X|T_s\geq t_s, T_u\geq t_s} \left[E\left[Y(t_s)|T_s=t_s, T_u(\infty)\geq t_s, X\right]\right] =$$
$$E_{X|T_s\geq t_s, T_u\geq t_s} \left[E\left[Y(t_s)|T_s\geq t_s, T_u(\infty)\geq t_s, X\right]\right] =$$
$$E_{X|T_s\geq t_s, T_u\geq t_s} \left[E\left[Y(t_s)|T_s\geq t_s, T_u(\infty)\geq t_s, X\right]\right] =$$

where the first three equalities follow by re-writing, the fourth by substituting for $w_{ATE1}^{t_s}(X) = \frac{1}{p(t_s,X)}$, the fifth by Assumptions 1 and 2 and the observational rule, the sixth equality by Assumption 3 for period t_s , and the seventh by averaging over X.

For the second part of $ATE(t_s)$, the estimator without the normalization is:

$$\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}\geq t_s}w_{ATE0}^{t_s}(T_{u,i},X_i)Y_i,$$

using similar reasoning as in (A.2) we have

$$E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}\geq t_s}w_{ATE0}^{t_s}(T_{u,i},X_i)Y_i\right] =$$
(A.11)

$$E_{X|T_s \ge t_s, T_u \ge t_s} \left[E \left[\sum_{t_u=t_s}^{T_u^{\max}} w_{ATE0}^{t_s}(t_u, X) \mathbf{I}(T_s > t_u, T_u = t_u) Y | T_s \ge t_s, T_u \ge t_s, X \right] \right].$$

Under Assumptions 1–3, and using the fact that $w_{ATE0}^{t_s}(t_u, X) = \frac{1}{\prod_{m=t_s}^{t_u} [1 - p(m, X)]}$:

$$E\left[w_{ATE0}^{t_{s}}(t_{u},X)\mathbf{I}(T_{s} > t_{u},T_{u} = t_{u})Y|T_{s} \ge t_{s}, T_{u} \ge t_{s},X\right] =$$
(A.12)
$$w_{ATE0}^{t_{s}}(t_{u},X)\Pr(T_{s} > t_{u},T_{u} = t_{u}|T_{u} \ge t_{s}, T_{s} \ge t_{s},X)E[Y|T_{s} > t_{u},T_{u} = t_{u},X] =$$

$$\frac{1}{\prod_{m=t_s}^{t_u} [1-p(m|X)]} h(t_u, X) \prod_{m=t_s}^{t_u-1} [1-h(m, X)] \prod_{m=t_s}^{t_u} [1-p(m, X)] E[Y|T_s > t_u, T_u = t_u, X] = h(t_u, X) \prod_{m=t_s}^{t_u-1} [1-h(m, X)] E[Y|T_s > t_u, T_u = t_u, X] = h(t_u, X) \prod_{m=t_s}^{t_u-1} [1-h(m, X)] E[Y(\infty)|T_s > t_u, T_u(\infty) = t_u, X] = h(t_u, X) \prod_{m=t_s}^{t_u-1} [1-h(m, X)] E[Y(\infty)|T_s \ge t_s, T_u(\infty) = t_u, X].$$

Note that the second equality follows from the definition of $w_{ATE0}^{t_s}(t_u, X)$, the third equality by simplifying, the fourth equality by Assumptions 1 and 2, and the fifth equality by applying Assumption 3 for $t_s, ..., t_u$.

Thus, from (A.11) and (A.12)

$$E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}, T_{u,i}\geq t_s} w_{ATE0}^{t_s}(T_{u,i}, X_i)Y_i\right] =$$
(A.13)

$$E_{X|T_{s}\geq t_{s},T_{u}\geq t_{s}}\left[\sum_{t_{u}=t_{s}}^{T_{u}^{\max}}h(t_{u},X)\prod_{m=t_{s}}^{t_{u}-1}[1-h(m,X)]E[Y(\infty)|T_{s}\geq t_{s},T_{u}(\infty)=t_{u},X]\right].$$

Next, using similar reasoning as for (A.8) we have

$$\sum_{t_u=t_s}^{T_u^{\max}} h(t_u, X) \prod_{m=t_s}^{t_u-1} [1 - h(m, X)] E[Y(\infty) | T_s \ge t_s, T_u(\infty) = t_u, X] =$$
(A.14)
$$E[Y(\infty) | T_s \ge t_s, T_u(\infty) \ge t_s, X],$$

so that from (A.13) and (A.14)

$$E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}, T_{u,i}\geq t_s} w_{ATE0}^{t_s}(T_{u,i}, X_i)Y_i\right] =$$

$$E_{X|T_s\geq t_s, T_u\geq t_s}\left[E[Y(\infty)|T_s\geq t_s, T_u(\infty)\geq t_s, X]\right] =$$

$$E[Y(\infty)|T_s\geq t_s, T_u(\infty)\geq t_s].$$
(A.15)

Finally, (A.10) and (A.15) imply that $E\left[\widehat{ATE}(t_s)\right] = ATE(t_s)$.

Estimation of $ATET(t_s)$ with short-run outcomes

Consider estimation of $ATET(t_s, \tau)$ and the estimator in (34)

$$\operatorname{ATET}(t_s, \tau) = E(Y_{t_s+\tau}(t_s) - Y_{t_s+\tau}(\infty) | T_s = t_s, T_u(t_s) \ge t_s).$$

For the first part of ATET(t_s , τ), the estimator without the normalization is:

$$\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}=t_s,T_{u,i}\geq t_s}Y_{t_s+\tau,i}.$$

By similar reasoning as for (A.1) we have

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}=t_s}Y_{t_s+\tau,i}\right] = E\left[Y_{t_s+\tau}(t_s)|T_s=t_s, T_u(t_s)\ge t_s\right].$$
(A.16)

For the second part of ATET(t_s , τ) the estimator without the normalization is:

$$\frac{1}{\rho_{t_s}N_{t_s}}\left[\sum_{i\in T_{s,i}>T_{u,i},t_s+\tau\geq T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)Y_{t_s+\tau,i}+\sum_{i\in T_{s,i}>t_s+\tau,T_{u,i}>t_s+\tau}w^{t_s}_{\tau}(T_{u,i},X_i)Y_{t_s+\tau,i}\right].$$

Initially, using similar reasoning as for (A.2)

$$\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i},t_s+\tau\geq T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i)Y_{t_s+\tau,i}\right] =$$
(A.17)
$$\left[\frac{1}{2}E\left[\sum_{i=1}^{t_s+\tau}w^{t_s}(t_i,X)\mathbf{I}(T_i>t_i,T_i=t_i)Y_{t_s+\tau}|T_i>t_i,X_i\right]\right]$$

$$E_{X|T_s \geq t_s, T_u \geq t_s} \left\lfloor \frac{1}{\rho_{t_s}} E\left\lfloor \sum_{t_u = t_s} w^{t_s}(t_u, X) \mathbf{I}(T_s > t_u, T_u = t_u) Y_{t_s + \tau} | T_s \geq t_s, T_u \geq t_s, X \right\rfloor \right\rfloor,$$

and using similar reasoning as for (A.3), we have that if Assumptions 1 and 3 hold, then

$$E\left[w^{t_s}(t_u, X)\mathbf{I}(T_s > t_u, T_u = t_u)Y_{t_s + \tau} | T_s \ge t_s, T_u \ge t_s, X\right] =$$
(A.18)

$$p(t_s, X)h(t_u, X)\prod_{m=t_s}^{t_u-1} [1-h(m, X)]E[Y_{t_s+\tau}(\infty)|T_s=t_s, T_u(\infty)=t_u, X].$$

Next,

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}>t_s+\tau, T_{u,i}>t_s+\tau}w_{\tau}^{t_s}(T_{u,i},X_i)Y_{t_s+\tau,i}\right] =$$
(A.19)

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}\geq t_s, T_{u,i}\geq t_s}w_{\tau}^{t_s}(T_{u,i},X_i)\mathbf{I}(T_{s,i}>t_s+\tau, T_{u,i}>t_s+\tau)Y_{t_s+\tau,i}\right] =$$

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}\geq t_s, T_{u,i}\geq t_s}\sum_{t_u>t_s+\tau}^{T_u^{max}}w_{\tau}^{t_s}(t_u,X_i)\mathbf{I}(T_{s,i}>t_s+\tau, T_{u,i}=t_u)Y_{t_s+\tau,i}\right] =$$

$$\frac{1}{\rho_{t_s}}E\left[\sum_{t_u>t_s+\tau}^{T_u^{max}}w_{\tau}^{t_s}(t_u,X)\mathbf{I}(T_s>t_s+\tau, T_u=t_u)Y_{t_s+\tau}|T_s\geq t_s, T_u\geq t_s\right] =$$

$$E_{X|T_s\geq t_s, T_u\geq t_s}\left[\frac{1}{\rho_{t_s}}E\left[\sum_{t_u>t_s+\tau}^{T_u^{max}}w_{\tau}^{t_s}(t_u,X)\mathbf{I}(T_s>t_s+\tau, T_u=t_u)Y_{t_s+\tau}|T_s\geq t_s, T_u\geq t_s, X\right]\right],$$

Then, Assumptions 1 and 3 and the fact that $w_{\tau}^{t_s}(X) = \frac{p(t_s, X)}{\prod_{m=t_s}^{t_s+\tau}[1-p(m, X)]}$ imply that

$$E\left[w_{\tau}^{t_{s}}(X)\mathbf{I}(T_{s} > t_{s} + \tau, T_{u} = t_{u})Y_{t_{s}+\tau}|T_{s} \ge t_{s}, T_{u} \ge t_{s}, X\right] =$$
(A.20)

$$w_{\tau}^{t_{s}}(X)\Pr(T_{s} > t_{s} + \tau, T_{u} = t_{u}|T_{u} \ge t_{s}, T_{s} \ge t_{s}, X)E[Y_{t_{s}+\tau}|T_{s} > t_{s} + \tau, T_{u} = t_{u}, X] =$$

$$\frac{p(t_{s}, X)}{\prod_{m=t_{s}}^{t_{s}+\tau}[1 - p(m, X)]}h(t_{u}, X)\prod_{m=t_{s}}^{t_{u}-1}[1 - h(m, X)]\prod_{m=t_{s}}^{t_{s}+\tau}[1 - p(m, X)]E[Y_{t_{s}+\tau}|T_{s} > t_{s} + \tau, T_{u} = t_{u}, X] =$$

$$p(t_{s}, X)h(t_{u}, X)\prod_{m=t_{s}}^{t_{u}-1}[1 - h(m, X)]E[Y_{t_{s}+\tau}|T_{s} > t_{s} + \tau, T_{u} = t_{u}, X] =$$

$$p(t_{s}, X)h(t_{u}, X)\prod_{m=t_{s}}^{t_{u}-1}[1 - h(m, X)]E[Y_{t_{s}+\tau}(\infty)|T_{s} > t_{s} + \tau, T_{u}(\infty) = t_{u}, X] =$$

$$p(t_{s}, X)h(t_{u}, X)\prod_{m=t_{s}}^{t_{u}-1}[1 - h(m, X)]E[Y_{t_{s}+\tau}(\infty)|T_{s} = t_{s}, T_{u}(\infty) = t_{u}, X] =$$

From (A.17)-(A.20) we have

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\left[\sum_{\substack{i\in T_{s,i}>T_{u,i},\\t_s+\tau\geq T_{u,i}\geq t_s}} w^{t_s}(T_{u,i},X_i)Y_{t_s+\tau,i} + \sum_{\substack{i\in T_{s,i}>t_s+\tau,\\T_{u,i}>t_s+\tau}} w^{t_s}(T_{u,i},X_i)Y_{t_s+\tau,i}\right]\right] = (A.21)$$

$$E_{X|T_{s}\geq t_{s},T_{u}\geq t_{s}}\left[\frac{p(t_{s},X)}{\rho_{t_{s}}}\sum_{t_{u}=t_{s}}^{T_{u}^{\max}}h(t_{u},X)\prod_{m=t_{s}}^{t_{u}-1}[1-h(m,X)]E[Y_{t_{s}+\tau}(\infty)|T_{s}=t_{s},T_{u}(\infty)=t_{u},X]\right].$$

Then, using similar reasoning as for (A.8)

$$\sum_{t_u=t_s}^{T_u^{\max}} h(t_u, X) \prod_{m=t_s}^{t_u-1} [1 - h(m, X)] E[Y_{t_s+\tau}(\infty) | T_s = t_s, T_u(\infty) = t_u, X] =$$
(A.22)
$$E[Y_{t_s+\tau}(\infty) | T_s = t_s, T_u(\infty) \ge t, X],$$

and, thus from (A.21) and (A.22)

$$E\left[\frac{1}{\rho_{t_{s}}N_{t_{s}}}\left[\sum_{\substack{i\in T_{s,i}>T_{u,i,}\\t_{s}+\tau\geq T_{u,i}\geq t_{s}}}w^{t_{s}}(T_{u,i},X_{i})Y_{t_{s}+\tau,i}+\sum_{\substack{i\in T_{s,i}>t_{s}+\tau,\\T_{u,i}>t_{s}+\tau}}w^{t_{s}}_{\tau}(T_{u,i},X_{i})Y_{t_{s}+\tau,i}\right]\right]= (A.23)$$

$$\frac{1}{\rho_{t_{s}}}E_{X|T_{s}\geq t_{s},T_{u}\geq t_{s}}\left[p(t_{s},X)E[Y_{t_{s}+\tau}(\infty)|T_{s}=t_{s},T_{u}(\infty)\geq t_{s},X]\right]=$$

$$\frac{1}{\rho_{t_{s}}}E_{X|T_{s}\geq t_{s},T_{u}\geq t_{s}}\left[\Pr(T_{s}=t_{s}|T_{s}\geq t_{s},T_{u}\geq t_{s},X)E[Y_{t_{s}+\tau}(\infty)|T_{s}=t_{s},T_{u}(\infty)\geq t_{s},X]\right]=$$

$$\frac{1}{\rho_{t_{s}}}\Pr(T_{s}=t_{s}|T_{s}\geq t_{s},T_{u}\geq t_{s})E[Y_{t_{s}+\tau}(\infty)|T_{s}=t_{s},T_{u}(\infty)\geq t_{s}]=$$

$$E[Y_{t_{s}+\tau}(\infty)|T_{s}=t_{s},T_{u}(\infty)\geq t_{s}].$$

Finally, (A.16) and (A.23) imply that $E\left[\widehat{\text{ATET}}(t_s)\right] = \text{ATET}(t_s)$.

Identification with time-varying covariates

Consider identification of $ATET(t_s) = E(Y(t_s) - Y(\infty) | T_s = t_s, T_u(t_s) \ge t_s)$. For the first component, as before

$$E(Y(t_s)|T_s = t_s, T_u(t_s) \ge t_s) = E(Y|T_s = t_s, T_u \ge t_s).$$
(A.24)

For the second component, by Assumptions 1 and 2, and averaging over $X_{t_s}^-$

$$E(Y(\infty)|T_{s} = t_{s}, T_{u}(t_{s}) \ge t_{s}) = E(Y(\infty)|T_{s} = t_{s}, T_{u}(\infty) \ge t_{s}) =$$
(A.25)
$$= E_{X_{t_{s}}^{-}|T_{s} = t_{s}, T_{u} \ge t_{s}}[E(Y(\infty)|T_{s} = t_{s}, T_{u}(\infty) \ge t_{s}, X_{t_{s}}^{-})].$$

Then, by Assumption 6 for period t_s ,

$$E(Y(\infty)|T_s=t_s,T_u(\infty)\geq t_s,X_{t_s})=E(Y(\infty)|T_s>t_s,T_u(\infty)\geq t_s,X_{t_s})$$

Next, using the notation $h(X_t^-) = \Pr(T_u = t | T_s > t, T_u \ge t, X_t^-)$ we have

$$E(Y(\infty)|T_{s} > t_{s}, T_{u}(\infty) \ge t_{s}, X_{t_{s}}^{-}) = h(X_{t_{s}}^{-})E(Y(\infty)|T_{s} > t_{s}, T_{u}(\infty) = t_{s}, X_{t_{s}}^{-}) +$$
(A.26)
$$[1 - h(X_{t_{s}}^{-})]E(Y(\infty)|T_{s} > t_{s}, T_{u}(\infty) > t_{s}, X_{t_{s}}^{-}) =$$
$$h(X_{t_{s}}^{-})E(Y|T_{s} > t_{s}, T_{u} = t_{s}, X_{t_{s}}^{-}) + [1 - h(X_{t_{s}}^{-})]E(Y(\infty)|T_{s} > t_{s}, T_{u}(\infty) > t_{s}, X_{t_{s}}^{-}),$$

where the second equality follows from Assumption 2 and the observational rule. Note that we also use that under Assumption 1 $\Pr(T_u = t | T_s > t, T_u(\infty) \ge t, X_t^-) = \Pr(T_u = t | T_s > t, T_u \ge t, X_t^-) = h(X_t^-)$, and the treatment probability $h(X_{t_s}^-)$ is observed. Next,

$$E(Y(\infty)|T_{s} > t_{s}, T_{u}(\infty) > t_{s}, X_{t_{s}}^{-}) =$$

$$E_{X_{t_{s}+1}^{-}|T_{s}>t_{s}, T_{u}>t_{s}, X_{t_{s}}^{-}}[E(Y(\infty)|T_{s} > t_{s}, T_{u}(\infty) > t_{s}, X_{t_{s}+1}^{-})] =$$

$$E_{X_{t_{s}+1}^{-}|T_{s}>t_{s}, T_{u}>t_{s}, X_{t_{s}}^{-}}[E(Y(\infty)|T_{s} > t_{s}+1, T_{u}(\infty) > t_{s}, X_{t_{s}+1}^{-})] =$$

$$E_{X_{t_{s}+1}^{-}|T_{s}>t_{s}, T_{u}>t_{s}, X_{t_{s}}^{-}}[E(Y(\infty)|T_{s} > t_{s}+1, T_{u}(\infty) > t_{s}+1, X_{t_{s}+1}^{-})] =$$

where the first equality follows from the law of iterated expectations, the second equality from Assumption 6 for period $t_s + 1$, and the third equality by rewriting. Here, the covariates $X_{t_s+1}^-$ may include $X_{t_s}^-$. From (A.26), by replacing t_s with $t_s + 1$:

$$E(Y(\infty)|T_s > t_s + 1, T_u(\infty) \ge t_s + 1, X_{t_s+1}^-) =$$
(A.28)

$$h(X_{t_{s+1}}^{-})E(Y(\infty)|T_s > t_s + 1, T_u(\infty) = t_s + 1, X_{t_s+1}^{-}) + [1 - h(X_{t_{s+1}}^{-})]E(Y(\infty)|T_s > t_s + 1, T_u(\infty) > t_s + 1, X_{t_s+1}^{-}).$$

Next, from (A.26) and (A.28)

$$\begin{split} E(Y(\infty)|T_s > t_s, T_u(\infty) \ge t_s, X_{t_s}^-) &= h(X_{t_s}^-) E(Y|T_s > t_s, T_u = t_s, X_{t_s}^-) + \\ [1 - h(X_{t_s}^-)] E_{X_{t_s+1}^-|T_s > t_s, T_u > t_s, X_{t_s}^-} [h(X_{t_s+1}^-) E(Y|T_s > t_s + 1, T_u = t_s + 1, X_{t_s+1}^-) + \\ [1 - h(X_{t_s+1}^-)] E(Y(\infty)|T_s > t_s + 1, T_u(\infty) > t_s + 1, X_{t_s+1}^-)]. \end{split}$$

Then, using (A.27) for $t_s + 1$ gives

$$E(Y(\infty)|T_{s} > t_{s}, T_{u} \ge t_{s}, X_{t_{s}}^{-}) = h(X_{t_{s}}^{-})E(Y|T_{s} > t_{s}, T_{u} = t_{s}, X_{t_{s}}^{-}) +$$

$$[1 - h(X_{t_{s}}^{-})]E_{X_{t_{s}+1}^{-}|T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}}[h(X_{t_{s}+1}^{-})E(Y|T_{s} > t_{s}+1, T_{u} = t_{s}+1, X_{t_{s}+1}^{-}) +$$

$$[1 - h(X_{t_{s+1}}^{-})]E_{X_{t_{s}+2}^{-}|T_{s} > t_{s}+1, T_{u} > t_{s}+1, X_{t_{s}+1}^{-}}[E(Y(\infty)|T_{s} > t_{s}+2, T_{u}(\infty) \ge t_{s}+2, X_{t_{s}+2}^{-})]],$$

and (A.26) for $t_s + 2$ gives

$$\begin{split} E(Y(\infty)|T_{s} > t_{s}, T_{u} \geq t_{s}, X_{t_{s}}^{-}) &= h(X_{t_{s}}^{-})E(Y|T_{s} > t_{s}, T_{u} = t_{s}, X_{t_{s}}^{-}) + \\ & [1 - h(X_{t_{s}}^{-})]E_{X_{t_{s}+1}^{-}|T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}}[h(X_{t_{s}+1}^{-})E(Y|T_{s} > t_{s}+1, T_{u} = t_{s}+1, X_{t_{s}+1}^{-}) + \\ & [1 - h(X_{t_{s+1}}^{-})]E_{X_{t_{s}+2}^{-}|T_{s} > t_{s}+1, T_{u} > t_{s}+1, X_{t_{s}+1}^{-}}[h(X_{t_{s+2}}^{-})E(Y|T_{s} > t_{s}+2, T_{u} = t_{s}+2, X_{t_{s}+2}^{-}) + \\ & [1 - h(X_{t_{s+2}}^{-})]E(Y(\infty)|T_{s} > t_{s}+2, T_{u}(\infty) > t_{s}+2, X_{t_{s}+2}^{-})]], \end{split}$$

and (A.27) for $t_s + 2$ gives

$$\begin{split} E(Y(\infty)|T_{s} > t_{s}, T_{u} \geq t_{s}, X_{t_{s}}^{-}) &= h(X_{t_{s}}^{-})E(Y|T_{s} > t_{s}, T_{u} = t_{s}, X_{t_{s}}^{-}) + \\ & [1 - h(X_{t_{s}}^{-})]E_{X_{t_{s}+1}^{-}|T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}}[h(X_{t_{s}+1}^{-})E(Y|T_{s} > t_{s}+1, T_{u} = t_{s}+1, X_{t_{s}+1}^{-}) + \\ & [1 - h(X_{t_{s+1}}^{-})]E_{X_{t_{s}+2}^{-}|T_{s} > t_{s}+1, T_{u} > t_{s}+1, X_{t_{s}+1}^{-}}[h(X_{t_{s+2}}^{-})E(Y|T_{s} > t_{s}+2, T_{u} = t_{s}+2, X_{t_{s}+2}^{-}) + \\ & [1 - h(X_{t_{s+2}}^{-})]E_{X_{t_{s}+3}^{-}|T_{s} > t_{s}+2, T_{u} > t_{s}+2, X_{t_{s}+2}^{-}}[E(Y(\infty)|T_{s} > t_{s}+3, T_{u}(\infty) \geq t_{s}+3, X_{t_{s}+1}^{-})]]]. \end{split}$$

and iteratively using (A.26) and (A.27) for $t_s + 3, \ldots, T_u^{\text{max}}$ we have

$$E(Y(\infty)|T_{s} > t_{s}, T_{u} \ge t_{s}, X_{t_{s}}^{-}) = h(X_{t_{s}}^{-})E(Y|T_{s} > t_{s}, T_{u} = t_{s}, X_{t_{s}}^{-}) +$$
(A.29)

$$[1 - h(X_{t_{s}}^{-})]E_{X_{t_{s}+1}^{-}|T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}}[h(X_{t_{s}+1}^{-})E(Y|T_{s} > t_{s} + 1, T_{u} = t_{s} + 1, X_{t_{s}+1}^{-}) +$$

$$[1 - h(X_{t_{s+1}}^{-})]E_{X_{t_{s}+2}^{-}|T_{s} > t_{s} + 1, T_{u} > t_{s} + 1, X_{t_{s}+1}^{-}}[h(X_{t_{s+2}}^{-})E(Y|T_{s} > t_{s} + 2, T_{u} = t_{s} + 2, X_{t_{s}+2}^{-}) + \dots +$$

$$[1 - h(X_{T_{u}^{-}}^{-})]E_{X_{T_{u}^{-}}^{-}max}|T_{s} > T_{u}^{max} - 1, T_{u} > T_{u}^{max} - 1, X_{T_{u}^{-}max}^{-}][p(X_{T_{u}^{-}}^{-})E(Y|T_{s} > T_{u}^{max}, T_{u} = T_{u}^{max}, X_{T_{u}^{-}}^{-})].\dots]].$$

Finally, combining (A.25) and (A.29) second component together with for (A.24) give the result in Theorem 4.

Estimation with time-varying covariates

We show that the if Assumptions 1, 2 and 6 hold, the estimator in (35), is an unbiased estimator of

$$ATET(t_s) = E(Y(t_s) - Y(\infty) \mid T_s = t_s, T_u(t_s) \ge t_s).$$

For the first part of $ATET(t_s)$, we have from (A.1)

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}=t_s,T_{u,i}\geq t_s}Y_i\right] = E\left[Y(t_s)|T_s=t_s,T_u(t_s)\geq t_s\right].$$
 (A.30)

For the second part of $ATET(t_s)$, the estimator without the normalization is:

$$\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i^-)Y_i,$$

using similar reasoning as for A.2 we have

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}\geq t_s}w^{t_s}(T_{u,i},X_i^-)Y_i\right] =$$
(A.31)

$$E_{X_{t_s}^-|T_s\geq t_s,T_u\geq t_s}\left[\frac{1}{\rho_{t_s}}E\left[\sum_{t_u=t_s}^{T_u^{\max}}w^{t_s}(t_u,X^-)\mathbf{I}(T_s>t_u,T_u=t_u)Y|T_s\geq t_s,T_u\geq t_s,X_{t_s}^-\right]\right].$$

We use the notation

$$h(t, X_t^-) = \Pr(T_u = t | T_s > t, T_u \ge t, X_t^-).$$

If Assumptions 1 and 2 and 6 hold, and since $w^{t_s}(t_u, X^-) = \frac{p(t_s, X_{t_s}^-)}{\prod_{m=t_s}^{t_u} [1-p(m, X_m^-)]}$:

$$E\left[w^{t_s}(t_s+1,X^{-})\mathbf{I}(T_s>t_s+1,T_u=t_s+1)Y|T_s\ge t_s,T_u\ge t_s,X_{t_s}^{-}\right] =$$
(A.32)

$$w^{t_s}(t_s + 1, X^-) \Pr(T_s > t_s, T_u > t_s | T_u \ge t_s, T_s \ge t_s, X_{t_s}^-) \times \\ \Pr(T_s > t_s + 1, T_u = t_s + 1 | T_s > t_s, T_u > t_s, X_{t_s}^-) E[Y|T_s > t_s + 1, T_u = t_s + 1, X_{t_s}^-] = \\$$

$$\frac{p(t_s, X_{t_s}^-)}{\prod_{m=t_s}^{t_s+1} [1 - p(m, X_m^-)]} [1 - h(t_s, X_{t_s}^-)] [1 - p(t_s, X_{t_s}^-)] \times \Pr(T_s > t_s + 1, T_u = t_s + 1 | T_s > t_s, T_u > t_s, X_{t_s}^-) E[Y|T_s > t_s + 1, T_u = t_s + 1, X_{t_s}^-] =$$

$$\frac{p(t_s, X_{t_s}^-)[1 - h(t_s, X_{t_s}^-)]}{[1 - p(t_s, X_{t_s+1}^-)]} \Pr(T_s > t_s + 1, T_u = t_s + 1 | T_s > t_s, T_u > t_s, X_{t_s}^-) \times E[Y|T_s > t_s + 1, T_u = t_s + 1, X_{t_s}^-].$$

Next,

$$\frac{\Pr(T_{s} > t_{s} + 1, T_{u} = t_{s} + 1 | T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-})}{1 - p(t_{s}, X_{t_{s}+1}^{-})} =$$
(A.33)

$$E_{X_{t_{s}+1}^{-}|T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}} \left[\frac{\Pr(T_{s} > t_{s} + 1, T_{u} = t_{s} + 1 | T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}+1}^{-})}{1 - p(t_{s}, X_{t_{s}+1}^{-})} \right] =$$

$$E_{X_{t_{s}+1}^{-}|T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}} \left[\frac{1 - p(t_{s}, X_{t_{s}+1}^{-}) \Pr(T_{u} = t_{s} + 1 | T_{s} > t_{s} + 1, T_{u} > t_{s}, X_{t_{s}+1}^{-})}{1 - p(t_{s}, X_{t_{s}+1}^{-})} \right] =$$

$$E_{X_{t_{s}+1}^{-}|T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}} \left[\Pr(T_{u} = t_{s} + 1 | T_{s} > t_{s} + 1, T_{u} > t_{s}, X_{t_{s}+1}^{-}) \right] =$$

$$E_{X_{t_{s}+1}^{-}|T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}} \left[\Pr(T_{u} = t_{s} + 1 | T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}+1}^{-}) \right] =$$

$$\Pr(T_{u} = t_{s} + 1 | T_{s} > t_{s}, T_{u} > t_{s}, X_{t_{s}}^{-}) = h(t_{s} + 1, X_{t_{s}}^{-}).$$

Note that the fourth equality follows from Assumption 6. Then, by (A.32) and (A.33), and using Assumptions 2 and 6

$$E\left[w^{t_{s}}(t_{s}+1,X^{-})\mathbf{I}(T_{s}>t_{s}+1,T_{u}=t_{s}+1)Y|T_{s}\geq t_{s},T_{u}\geq t_{s},X_{t_{s}}^{-}\right] =$$
(A.34)
$$p(t_{s},X_{t_{s}}^{-})h(t_{s}+1,X_{t_{s}}^{-})[1-h(t_{s},X_{t_{s}}^{-})]E[Y|T_{s}>t_{s}+1,T_{u}=t_{s}+1,X_{t_{s}}^{-}] =$$
$$p(t_{s},X_{t_{s}}^{-})h(t_{s}+1,X_{t_{s}}^{-})[1-h(t_{s},X_{t_{s}}^{-})]E[Y(\infty)|T_{s}>t_{s}+1,T_{u}(\infty)=t_{s}+1,X_{t_{s}}^{-}] =$$
$$p(t_{s},X_{t_{s}}^{-})h(t_{s}+1,X_{t_{s}}^{-})[1-h(t_{s},X_{t_{s}}^{-})]E[Y(\infty)|T_{s}=t_{s},T_{u}(\infty)=t_{s}+1,X_{t_{s}}^{-}],$$

where the second equality follows from Assumption 2 and the third by Assumption 6.

By similar reasoning as for (A.32)-(A.34) we have

$$E\left[w^{t_{s}}(t_{u},X^{-})\mathbf{I}(T_{s} > t_{u},T_{u} = t_{u})Y|T_{s} \ge t_{s}, T_{u} \ge t_{s}, X_{t_{s}}^{-}\right] =$$
(A.35)
$$p(t_{s},X_{t_{s}}^{-})h(t_{u},X_{t_{s}}^{-})\prod_{m=t_{s}}^{t_{u}-1}[1-h(m,X_{t_{s}}^{-})]E[Y(\infty)|T_{s} = t_{s},T_{u}(\infty) = t_{u},X_{t_{s}}^{-}].$$

Thus, from (A.31)-(A.35):

$$E\left[\frac{1}{\rho_{t_s}N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}\ge t_s}w^{t_s}(T_{u,i},X_i^-)Y_i\right] =$$
(A.36)

$$E_{X_{t_s}^-|T_s \ge t_s, T_u \ge t_s} \left[\frac{p(t_s, X_{t_s}^-)}{\rho_{t_s}} \sum_{t_u=t_s}^{T_u^{\text{max}}} h(t_u, X_{t_s}^-) \prod_{m=t_s}^{t_u-1} [1 - h(m, X_{t_s}^-)] E[Y(\infty)|T_s = t_s, T_u(\infty) = t_u, X_{t_s}^-] \right]$$

.

Next, by similar reasoning as for (A.5)-(A.8) we have

$$\sum_{t_{u}=t_{s}}^{T_{u}^{\max}} h(t_{u}, X_{t_{s}}^{-}) \prod_{m=t_{s}}^{t_{u}-1} [1 - h(m, X_{t_{s}}^{-})] E[Y(\infty)|T_{s} = t_{s}, T_{u}(\infty) = t_{u}, X_{t_{s}}^{-}] =$$
(A.37)
$$E[Y(\infty)|T_{s} = t_{s}, T_{u}(\infty) \ge t_{s}, X_{t_{s}}^{-}],$$

so that from (A.36) and (A.37)

$$E\left[\frac{1}{\rho_{t_{s}}N_{t_{s}}}\sum_{i\in T_{s,i}>T_{u,i}\geq t_{s}}w^{t_{s}}(T_{u,i},X_{i}^{-})Y_{i}\right] =$$

$$E_{X_{t_{s}}^{-}|T_{s}\geq t_{s},T_{u}\geq t_{s}}\left[\frac{p(t_{s},X_{t_{s}}^{-})}{\rho_{t_{s}}}E[Y(\infty)|T_{s}=t_{s},T_{u}(\infty)\geq t_{s},X_{t_{s}}^{-}]\right] =$$

$$\frac{1}{\rho_{t_{s}}}E_{X_{t_{s}}^{-}|T_{s}\geq t_{s},T_{u}\geq t_{s}}\left[\Pr(T_{s}=t_{s}|T_{s}\geq t_{s},T_{u}\geq t_{s},X_{t_{s}}^{-})E[Y(\infty)|T_{s}=t_{s},T_{u}(\infty)\geq t_{s},X_{t_{s}}^{-}]\right] =$$

$$\frac{1}{\rho_{t_{s}}}\Pr(T_{s}=t_{s}|T_{s}\geq t_{s},T_{u}\geq t_{s})E[Y(\infty)|T_{s}=t_{s},T_{u}(\infty)\geq t_{s}] =$$

$$E[Y(\infty)|T_{s}=t_{s},T_{u}(\infty)\geq t_{s}].$$
(A.38)

Finally, (A.30) and (A.38) imply that $E\left[\widehat{\text{ATET}}(t_s)\right] = \text{ATET}(t_s)$.

Identification with right-censored durations

Consider identification of $ATET(t_s) = E(Y(t_s) - Y(\infty) | T_s = t_s, T_c > t_s, T_u(t_s) \ge t_s)$ under Assumptions 1-3 and 7. First, consider $E(Y(\infty)|T_s = t_s, T_c > t_s, T_u(t_s) \ge t_s)$. Initially, by Assumptions 1 and 2 and the law of iterated expectations:

$$E(Y(\infty) \mid T_s = t_s, T_c > t_s, T_u(t_s) \ge t_s) = E(Y(\infty) \mid T_s = t_s, T_c > t_s, T_u(\infty) \ge t_s) = (A.39)$$

$$E_{X|T_s=t_s,T_c>t_s,T_u\geq t_s}\left[E(Y(\infty)\mid T_s=t_s,T_c>t_s,T_u(\infty)\geq t_s,X)\right],$$

where the averaging over X is possible given common support. Next, if Assumption 3 holds for period t_s we have

$$E(Y(\infty) \mid T_s = t_s, T_c > t_s, T_u(\infty) \ge t_s, X) = E(Y(\infty) \mid T_s > t_s, T_c > t_s, T_u(\infty) \ge t_s, X).$$

Then, by the law of iterated expectations

$$E(Y(\infty) \mid T_s > t_s, T_c > t_s, T_u(\infty) \ge t_s, X) =$$

$$Pr(T_u = t_s \mid T_s > t_s, T_c > t_s, T_u(\infty) \ge t_s, X) E(Y(\infty) \mid T_s > t_s, T_c > t_s, T_u(\infty) = t_s, X) +$$

$$Pr(T_u > t_s \mid T_s > t_s, T_c > t_s, T_u(\infty) \ge t_s, X) E(Y(\infty) \mid T_s > t_s, T_c > t_s, T_u(\infty) > t_s, X),$$

$$(A.40)$$

decomposing the counterfactual outcome under never treatment into average outcomes for individuals with $T_u = t_s$ and $T_u > t_s$. For the group with $T_u = t_s$ in (A.40), we have by Assumption 2

$$E(Y(\infty)|T_s > t_s, T_c > t_s, T_u = t_s, X) = E(Y|T_s > t_s, T_c > t_s, T_u = t_s, X),$$
(A.41)

and the probabilities $\Pr(T_u = t_s | T_s > t_s, T_c > t_s, T_u(\infty) \ge t_s, X)$ and $\Pr(T_u > t_s | T_s > t_s, T_c > t_s, T_u(\infty) \ge t_s, X)$ are also observed.

For the group, with $T_u > t_s$, in (A.40), we have

$$\begin{split} & E(Y(\infty)|T_s \ > \ t_s, T_c > t_s, T_u > t_s, X) = \\ & E(Y(\infty)|T_s \ > \ t_s, T_c > t_s + 1, T_u > t_s, X) = \\ & E(Y(\infty)|T_s \ > \ t_s + 1, T_c > t_s + 1, T_u > t_s, X) = \\ & E(Y(\infty)|T_s \ > \ t_s + 1, T_c > t_s + 1, T_u \ge t_s, X) = \\ \end{split}$$

where the first equality follows from Assumption 7 for period $t_s + 1$, the second from Assumption 3 for period $t_s + 1$ and the third equality by rewriting. Next, for sake of

presentation, let us introduce some auxiliary notation:

$$h_c(t,X) = \Pr(T_u = t \mid T_s > t, T_c > t, T_u \ge t, X).$$

Using this notation and using (A.40) by replacing t_s with $t_s + 1$ we have

$$E(Y(\infty) | T_s > t_s + 1, T_c > t_s + 1, T_u \ge t_s + 1, X) =$$

$$h_c(t_s + 1, X)E(Y(\infty)|T_s > t_s + 1, T_c > t_s + 1, T_u = t_s + 1, X) +$$

$$[1 - h_c(t_s + 1, X)]E(Y(\infty)|T_s > t_s + 1, T_c > t_s + 1, T_u > t_s + 1, X),$$
(A.42)

so that (A.40)-(A.42) give

$$E(Y(\infty)|T_s > t_s, T_c > t_s, T_u \ge t_s, X) = h_c(t_s, X)E(Y|T_s > t_s, T_c > t_s, T_u = t_s, X) + [1 - h_c(t_s, X)]h_c(t_s + 1, X)E(Y|T_s > t_s + 1, T_c > t_s + 1, T_u = t_s + 1, X) + [1 - h_c(t_s, X)][1 - h_c(t_s + 1, X)]E(Y(\infty)|T_s > t_s + 1, T_c > t_s + 1, T_u > t_s + 1, X).$$

Then, iteratively using (A.40) and (A.41) for $t_s + 2, ..., T_u^{\text{max}}$ we have

$$E(Y(\infty)|T_{s} > t_{s}, T_{c} > t_{s}, T_{u} \ge t_{s}, X) =$$

$$\sum_{k=t_{s}}^{T_{us}^{max}} h_{c}(k, X) \prod_{m=t_{s}}^{k-1} [1 - h_{c}(m, X)] E(Y|T_{s} > k, T_{c} > k, T_{u} = k, X).$$
(A.43)

Then, from (A.39)-(A.43)

$$E(Y(\infty) \mid T_s = t_s, T_c > t_s, T_u(t_s) \ge t_s, X) =$$
(A.44)

$$E_{X|T_s=t_s,T_c>t_s,T_u\geq t_s}\left[\sum_{k=t_s}^{T_{us}^{max}} h_c(k,X)\prod_{m=t_s}^{k-1} [1-h_c(m,X)]E(Y|T_s>k,T_c>k,T_u=k,X)\right].$$

Second, for $E(Y(t_s)|T_s = t_s, T_c > t_s, T_u(t_s) \ge t_s)$, Assumptions 1 and 2 and the law of

iterated expectations give

$$E(Y(t_{s}) \mid T_{s} = t_{s}, T_{c} > t_{s}, T_{u}(t_{s}) \ge t_{s}) = E(Y(t_{s}) \mid T_{s} = t_{s}, T_{c} > t_{s}, T_{u} \ge t_{s}) =$$
(A.45)
$$E_{X|T_{s} = t_{s}, T_{c} > t_{s}, T_{u} \ge t_{s}} [E(Y(t_{s}) \mid T_{s} = t_{s}, T_{c} > t_{s}, T_{u} \ge t_{s}, X)].$$

Then, by the law of iterated expectations

$$E(Y(t_{s}) | T_{s} = t_{s}, T_{c} > t_{s}, T_{u} \ge t_{s}, X) =$$
(A.46)

$$Pr(T_{u} = t_{s} | T_{s} = t_{s}, T_{c} > t_{s}, T_{u} \ge t_{s}, X) E(Y(t_{s}) | T_{s} = t_{s}, T_{c} > t_{s}, T_{u} = t_{s}, X) +$$

$$Pr(T_{u} > t_{s} | T_{s} = t_{s}, T_{c} > t_{s}, T_{u} \ge t_{s}, X) E(Y(t_{s}) | T_{s} = t_{s}, T_{c} > t_{s}, T_{u} > t_{s}, X),$$

as above decomposing the outcome of interest into average outcomes for individuals with $T_u = t_s$ and $T_u > t_s$.

Next,

$$E(Y(t_s)|T_s = t_s, T_c > t_s, T_u = t_s, X) = E(Y|T_s = t_s, T_c > t_s, T_u = t_s, X).$$
(A.47)

For sake of presentation, let us introduce some additional auxiliary notation:

$$h_{c1}(t, X, t_s) = \Pr(T_u = t \mid T_s = t_s, T_c > t, T_u \ge t, X).$$

Then, using this notation iteratively using (A.46) and (A.47) for $t_s + 2, \ldots, T_u^{\text{max}}$ we have

$$E(Y(t_s)|T_s = t_s, T_c > t_s, T_u \ge t_s, X) =$$
 (A.48)

$$\sum_{k=t_s}^{T_{us}^{\max}} h_{c1}(t,X,t_s) \prod_{m=t_s}^{k-1} [1 - h_{c1}(t,X,t_s)] E(Y|T_s = t_s, T_c > k, T_u = k, X),$$

so that by (A.45)-(A.48)

$$E(Y(t_s) \mid T_s = t_s, T_c > t_s, T_u(t_s) \ge t_s) =$$
(A.49)

$$E_{X|T_s=t_s,T_c>t_s,T_u\geq t_s}\left[\sum_{k=t_s}^{T_{us}^{\max}} h_{c1}(t,X,t_s)\prod_{m=t_s}^{k-1} [1-h_{c1}(t,X,t_s)]E(Y|T_s=t_s,T_c>k,T_u=k,X)\right]$$

Finally, (A.44) and (A.49) gives the result in Theorem 5.

Estimation with right-censored durations

We now show that if Assumptions 1-3 and 7 hold, the estimator in (37) is an unbiased estimator of $ATET(t_s)$. First, for the first component of $ATET(t_s)$, the estimator without the normalization is:

$$\frac{1}{\rho_{t_s}^c N_{t_s}^c} \sum_{i \in T_{s,i} = t_s, T_{c,i} > T_{u,i}, T_{u,i} \ge t_s} w_{c_1}^{t_s}(T_{u,i}, X_i) Y_i.$$

Using similar reasoning as above we have

$$E\left[\frac{1}{\rho_{t_{s}}^{c}N_{t_{s}}^{c}}\sum_{i\in T_{s,i}=t_{s},T_{c,i}>T_{u,i},T_{u,i}\geq t_{s}}w_{c_{1}}^{t_{s}}(T_{u,i},X_{i})Y_{i}\right] =$$
(A.50)

$$E\left[\frac{1}{\rho_{t_{s}}^{c}N_{t_{s}}^{c}}\sum_{i\in T_{s,i}=t_{s},T_{c,i}>t_{s},T_{u,i}\geq t_{s}}w_{c_{1}}^{t_{s}}(T_{u,i},X_{i})\mathbf{I}(T_{c,i}>T_{u,i})Y_{i}\right] =$$
$$E\left[w_{c_{1}}^{t_{s}}(T_{u},X)\mathbf{I}(T_{c}>T_{u})Y|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s}\right] =$$
$$E\left[\sum_{t_{u}=t_{s}}^{T_{u}^{max}}w_{c_{1}}^{t_{s}}(t_{u},X)\mathbf{I}(T_{c}>t_{u},T_{u}=t_{u})Y|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s}\right] =$$
$$E_{X|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s}}\left[E\left[\sum_{t_{u}=t_{s}}^{T_{u}^{max}}w_{c_{1}}^{t_{s}}(t_{u},X)\mathbf{I}(T_{c}>t_{u},T_{u}=t_{u})Y|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s},T_{u}\geq t_{s}$$

Introduce the notation

$$h_c(t,X) = \Pr(T_u = t | T_u \ge t, T_c > t, T_s > t, X).$$

Then, if Assumptions 1–3 and 7 hold, and noting that $w_{c_1}^{t_s}(t_u, X) = \frac{1}{\prod_{m=t_{s+1}}^{t_u} [1-e_{c_1}(m, t_s, X)]}$:

$$E\left[w_{c_{1}}^{t_{s}}(t_{u},X)\mathbf{I}(T_{c}>t_{u},T_{u}=t_{u})Y|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s},X\right] =$$
(A.51)

$$w_{C_1}^{t_s}(t_u, X) \Pr(T_c > t_u, T_u = t_u | T_u = t_s, T_c > t_s, T_s \ge t_s, X) \times E[Y | T_s = t_s, T_c > t_u, T_u = t_u, X] =$$

$$\begin{split} \frac{1}{\prod_{m=t_{s}+1}^{t_{u}}[1-e_{c1}(m,t_{s},X)]} & h_{c}(t_{u},X)\prod_{m=t_{s}}^{t_{u}-1}[1-h_{c}(m,X)]\prod_{m=t_{s}+1}^{t_{u}}[1-e_{c1}(m,t_{s},X)] \times \\ & E[Y|T_{s} = t_{u},T_{c} > t_{u},T_{u} = t_{u},X] = \\ & h_{c}(t_{u},X)\prod_{m=t_{s}}^{t_{u}-1}[1-h_{c}(m,X)]E[Y|T_{s} = t_{u},T_{c} > t_{u},T_{u} = t_{u},X] = \\ & h_{c}(t_{u},X)\prod_{m=t_{s}}^{t_{u}-1}[1-h_{c}(m,X)]E[Y(\infty)|T_{s} = t_{s},T_{c} > t_{u},T_{u}(\infty) = t_{u},X] = \\ & h_{c}(t_{u},X)\prod_{m=t_{s}}^{t_{u}-1}[1-h_{c}(m,X)]E[Y(\infty)|T_{s} = t_{s},T_{c} > t_{s},T_{u}(\infty) = t_{u},X] = \end{split}$$

where the second equality follows from the definition of $w_{c_0}^{t_s}(t_u, X)$, the third equality by simplifying, the fourth equality by Assumptions 2, and the fifth equality by applying Assumption 7 for $t_s, ..., t_u$.

From (A.50) and (A.51)

$$E\left[\frac{1}{\rho_{t_{s}}^{c}N_{t_{s}}^{c}}\sum_{i\in T_{s,i}=t_{s}, T_{c,i}>T_{u,i}, T_{u,i}\geq t_{s}}w_{c_{1}}^{t_{s}}(T_{u,i}, X_{i})Y_{i}\right] =$$
(A.52)

$$E_{X|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s}}\left[\sum_{t_{u}=t_{s}}^{T_{u}^{\max}}h_{c}(t_{u},X)\prod_{m=t_{s}}^{t_{u}-1}[1-h_{c}(m,X)]E[Y(\infty)|T_{s}=t_{s},T_{c}>t_{s},T_{u}(\infty)=t_{u},X]\right].$$

Next, by similar reasoning as for (A.8) we have

$$\sum_{t_u=t_s}^{T_u^M} h_c(t_u, X) \prod_{m=t_s}^{t_u-1} [1 - h_c(m, X)] E[Y(\infty)|T_s = t_s, T_c > t_s, T_u(\infty) = t_u, X] =$$
(A.53)
$$E[Y(\infty)|T_s = t_s, T_c > t_s, T_u(\infty) \ge t_s, X],$$

so that from (A.52) and (A.53)

$$E\left[\frac{1}{\rho_{t_{s}}^{c}N_{t_{s}}^{c}}\sum_{i\in T_{s,i}=t_{s},T_{c,i}>T_{u,i},T_{u,i}\geq t_{s}}w_{c_{1}}^{t_{s}}(T_{u,i},X_{i})Y_{i}\right] =$$

$$E_{X|T_{s}=t_{s},T_{c}>t_{s},T_{u}\geq t_{s}}\left[E[Y(\infty)|T_{s}=t_{s},T_{c}>t_{s},T_{u}(\infty)\geq t_{s},X]\right] =$$

$$E[Y(\infty)|T_{s}=t_{s},T_{c}>t_{s},T_{u}(\infty)\geq t_{s}].$$
(A.54)

Second, for the second component of $ATET(t_s)$ the estimator without the normalization is:

$$\frac{1}{N_{t_s}^c} \sum_{i \in T_{s,i} > T_{u,i}, T_{c,i} > T_{u,i}, T_{u,i} \ge t_s} w_{c_0}^{t_s}(T_{u,i}, X_i) Y_i.$$

Using similar reasoning as for (A.2) we have

$$E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}, T_{c,i}>T_{u,i}, T_{u,i}\geq t_s} w_{c_0}^{t_s}(T_{u,i}, X_i)Y_i\right] =$$
(A.55)

$$E_{X|T_s \ge t_s, T_c > t_s, T_u \ge t_s} \left[E \left[\sum_{t_u = t_s}^{T_u^{\max}} w_{c_0}^{t_s}(t_u, X) \mathbf{I}(T_s > t_u, T_c > t_u, T_u = t_u) Y | T_s \ge t_s, T_c > t_s, T_u \ge t_s, X \right] \right].$$

Next, if Assumptions 1–3 and 7 hold, and since

$$w_{c_0}^{t_s}(t_u, X) = \frac{p_c(t_s, X)}{\rho_{t_s}^c [1 - p_c(t_s, X)] \prod_{m=t_s+1}^{t_u} [1 - p_c(m, X)] [1 - e_{c_0}(m, X)]}$$

we have

$$E\left[w_{c_0}^{t_s}(t_u, X)\mathbf{I}(T_s > t_u, T_c > t_u, T_u = t_u)Y|T_s \ge t_s, T_c > t_s, T_u \ge t_s, X\right] =$$
(A.56)

$$w_{c_0}^{t_s}(t_u, X) \Pr(T_s > t_u, T_c > t_u, T_u = t_u | T_u \ge t_s, T_c > t_s, T_s \ge t_s, X) \times E[Y|T_s > t_u, T_c > t_u, T_u = t_u, X] =$$
$$\begin{split} &\frac{p_c(t_s,X)}{\rho_{t_s}^c[1-p_c(t_s,X)]\prod_{m=t_s+1}^{t_u}[1-p_c(m,X)][1-e_{c0}(m,X)]}h_c(t_u,X)\prod_{m=t_s}^{t_u-1}[1-h_c(m,X)]\times\\ &[1-p_c(t_s,X)]\prod_{m=t_s+1}^{t_u}[1-p(m,X)][1-e_{c0}(m,X)]E[Y|T_s>t_u,T_c>t_u,T_u=t_u,X]=\\ &\frac{p_c(t_s,X)}{\rho_{t_s}^c}h_c(t_u,X)\prod_{m=t_s}^{t_u-1}[1-h_c(m,X)]E[Y|T_s>t_u,T_c>t_u,T_u=t_u,X]=\\ &\frac{p_c(t_s,X)}{\rho_{t_s}^c}h_c(t_u,X)\prod_{m=t_s}^{t_u-1}[1-h_c(m,X)]E[Y(\infty)|T_s>t_u,T_c>t_u,T_u(\infty)=t_u,X]=\\ &\frac{p_c(t_s,X)}{\rho_{t_s}^c}h_c(t_u,X)\prod_{m=t_s}^{t_u-1}[1-h_c(m,X)]E[Y(\infty)|T_s=t_s,T_c>t_u,T_u(\infty)=t_u,X]=\\ &\frac{p_c(t_s,X)}{\rho_{t_s}^c}h_c(t_u,X)\prod_{m=t_s}^{t_u-1}[1-h_c(m,X)]E[Y(\infty)|T_s=t_s,T_c>t_s,T_u(\infty)=t_u,X]= \end{split}$$

where the second equality follows from the definition of $w_{c_0}^{t_s}(t_u, X)$, the third equality by simplifying, the fourth equality by Assumptions 2, the fifth equality by applying Assumption 3 for $t_s, ..., t_u$, and the sixth equality by applying Assumption 7 for $t_s, ..., t_u$.

From (A.55) and (A.56)

$$E\left[\frac{1}{N_{t_s}}\sum_{i\in T_{s,i}>T_{u,i}, T_{c,i}>T_{u,i}, T_{u,i}\geq t} w_{c0}^{t_s}(T_{u,i}, X_i)Y_i\right] =$$
(A.57)

$$E_{X|T_{s} \ge t_{s}, T_{c} > t_{s}, T_{u} \ge t_{s}} \left[\frac{p_{c}(t_{s}, X)}{\rho_{t_{s}}^{c}} \sum_{t_{u}=t_{s}}^{T_{u}^{\max}} h(t_{u}, X) \prod_{m=t_{s}}^{t_{u}-1} [1 - h(m, X)] \times .$$

$$E[Y(\infty)|T_{s} = t_{s}, T_{c} > t_{s}, T_{u}(\infty) = t_{u}, X].$$

Next, by similar reasoning as for (A.8) we have

$$\sum_{t_u=t_s}^{T_u^{\max}} h(t_u, X) \prod_{m=t_s}^{t_u-1} [1 - h(m, X)] E[Y(\infty) | T_s = t_s, T_c > t_s, T_u(\infty) = t_u, X] =$$
(A.58)
$$E[Y(\infty) | T_s = t_s, T_c > t_s, T_u(\infty) \ge t_s, X].$$

Thus, from (A.57) and (A.58)

$$E\left[\frac{1}{N_{t_{s}}}\sum_{i\in T_{s,i}>T_{u,i}, T_{c,i}>T_{u,i}, T_{u,i}\geq t} w_{c0}^{t_{s}}(T_{u,i}, X_{i})Y_{i}\right] =$$
(A.59)

$$E_{X|T_{s}\geq t_{s}, T_{c}>t_{s}, T_{u}\geq t_{s}}\left[\frac{p_{c}(t_{s}, X)}{\rho_{t_{s}}^{c}}E[Y(\infty)|T_{s}=t_{s}, T_{c}>t_{s}, T_{u}(\infty)\geq t_{s}, X]\right] =$$

$$\frac{\Pr(T_{s}=t_{s}|T_{u}\geq t_{s}, T_{c}>t_{s}, T_{s}\geq t_{s})}{\rho_{t_{s}}^{c}}E[Y(\infty)|T_{s}=t_{s}, T_{c}>t_{s}, T_{u}(\infty)\geq t_{s}] =$$

$$E[Y(\infty)|T_{s}=t_{s}, T_{c}>t_{s}, T_{u}(\infty)\geq t_{s}],$$

since $\rho_{t_s}^c = \Pr(T_s = t_s | T_u \ge t_s, T_c > t_s, T_s \ge t_s)$. Finally, (A.54) and (A.59) imply that $E\left[\widehat{\operatorname{ATET}}(t_s)\right] = \operatorname{ATET}(t_s)$.

Appendix B. Monte Carlo simulation

Simulation design

This simulation study examines properties of the estimator introduced in the paper. We use the following notation for the conditional exit probability out of the initial state: $\theta_{T_u}(t) = \Pr(T_u = t_u | T_u \ge t_u)$, and the conditional treatment probability: $\theta_{T_s}(t) = \Pr(T_s = t_s | T_u \ge t_s, T_s \ge t_s)$. We consider the following discrete time DGP:

$$\begin{aligned} \theta_{T_u}(t) &= f(-2.5 + \beta_u X + v_u) \quad (B.1) \\ \theta_{T_s}(t) &= f(\alpha_s + \beta_s X + v_s) \quad , t \le 12 \\ \theta_{T_s}(t) &= 0 \quad , t > 12 \\ Y &= 100 + \beta_y X + \delta I(T_s \ge T_u) + \beta_{v_u} v_u + v_y \\ & \text{with } X, v_u, v_s \sim \text{unif}(-1, 1), v_y \sim N(0, 5), \end{aligned}$$

with X, v_u, v_s, v_y all independently distributed of each other, and $f(h) = [1 + \exp(-h)]^{-1}$, i.e. we use a logistic model for the conditional exit and treatment probabilities.

This model has several properties worth noticing. First, the treatment can start at any point during the first 12 time periods, corresponding to a treatment in place during the first year (if the time period is a month). Second, both durations, T_u and T_s , and the outcome, Y, depend on observed and unobserved characteristics. However, since the unobserved effect in the treatment equation is uncorrelated with the other unobserved effects, the unconfoundedness assumption holds. Third, the unobserved effect in the duration time equation also appears in the long-run outcome equation. This is consistent with the idea that some unobserved characteristics determine both time in the initial state and the long-run outcome equation. In the training for unemployed example this may be unobserved motivation and/or unobserved ability.

In the baseline setting, the correlation between the unobserved characteristics in the exit and long-outcome equations β_{v_u} is 1, the baseline treatment probability parameter α_s is -3.0, the impact of the covariate on treatment β_s is 1, the treatment effect on the long-run outcome δ is 0, and impact of the covariate on the long-run outcomes β_Y is set

to 1. These parameters are then varied in four different ways. Model A varies the baseline treatment parameter (α_s between -4.5 and -1.5), and Model B varies the impact of the covariate on treatment (β_s between 0 and 2). With $\alpha_s = -4$, the conditional treatment probability in each period is 0.021 while with $\alpha_s = -2$ this is 0.13. If β_s equals 0.5 the conditional treatment probability varies between 0.029 and 0.076; and if $\beta_s = 1.5$, this probability varies between 0.011 and 0.18. Model C varies the correlation between the unobserved characteristics in the exit and long-outcome equations (β_u between 0 and 2). Finally, Model D allows the treatment effect on the long-run outcome, δ , to vary between 1 and 10.

We focus on the aggregated effect ATET. All propensity scores are estimated with a correct logistic model specification. We initially study the bias of each estimator. The sample size is set to 10,000 and the number of replications is 2000. Common support is imposed through the above-mentioned variant of the three-step approach from Huber et al. (2013), with the upper limit on the weight given to a certain observation set to 1%. After this we study the size and variance of the dynamic estimator, using bootstrapped standard errors (99 replications).

Simulation results

We compare the dynamic IPW estimator and a static IPW estimator. *Figure B.1* reports how the bias of the two estimators are related to the baseline treatment rate. As expected, the static IPW estimator is biased, and the bias is increasing in the treatment parameter (higher α_s). This is because a higher conditional treatment probability implies more extensive dynamic treatment assignment. The bias of the dynamic IPW estimator, on the other hand, is virtually zero for all treatment probabilities and roughly 100 times smaller than for the static IPW estimator.

Figure B.2 also shows that the bias of the static IPW estimator increases with the variance of the treatment probability across units (larger β_s), while the dynamic approach is unbiased for all values of β_s . From Figure B.3, it can also be seen that the bias of the static approach is increasing in the correlation between the unobserved characteristics in the exit and long-outcome equations, β_u . Again, the bias of our dynamic approach is

virtually zero.

Finally, *Table B.1* presents the bias, variance and size of our dynamic IPW estimator. The simulation results are for sample sizes of 1000 and 4000. We vary the parameters of the DGP in a similar was for Models A-D, but we only report simulation results for the baseline case and one additional case for each model. First, as expected based on the results in *Figure B.1–Figure B.4*, the bias is small in all cases. Size is for a test with nominal size of 5%, so that the IPW estimator roughly has correct size (Columns 3 and 6). The tables also shows that standard error decreases by roughly 50% when the sample size is increased by a factor of four from 1000 to 4000, suggesting that the estimator is \sqrt{N} -convergent.

Figure B.1: Simulated bias for our dynamic IPW estimator and a static IPW estimator. Model A: baseline treatment rate



Note: α_s is the conditional treatment probability parameter. The data generating processes for the logistic simulation models described in the text. Bias for aggregated effect of treatment over the first 12 months. Dynamic IPW is the estimator introduced in this paper. Static IPW is a standard static IPW estimator with normalized weights. Results are based on 2,000 replications.

Figure B.2: Simulated bias for our dynamic IPW estimator and a static IPW estimator. Model B: impact of the covariate on the conditional treatment probability



Note: β_s is the impact of the covariate on treatment. The data generating processes for the logistic simulation models described in the text. Bias for aggregated effect of treatment over the first 12 months. Dynamic IPW is the estimator introduced in this paper. Static IPW is a standard static IPW estimator with normalized weights. Results are based on 2,000 replications.

Figure B.3: Simulated bias for our dynamic IPW estimator and a static IPW estimator. Model C: correlation between the unobserved characteristics in the exit and long-outcome equations



Note: β_u determines the correlation between the unobserved characteristics in the exit and long-outcome equations. The data generating processes for the logistic simulation models described in the text. Bias for aggregated effect of treatment over the first 12 months. Dynamic IPW is the estimator introduced in this paper. Static IPW is a standard static IPW estimator with normalized weights. Results are based on 2,000 replications. **Figure B.4:** Simulated bias for our dynamic IPW estimator and a static IPW estimator. Model D: treatment effect on the long-run outcome



Note: δ is the treatment effect. The data generating processes for the logistic simulation models described in the text. Bias for aggregated effect of treatment over the first 12 months. Dynamic IPW is the estimator introduced in this paper. Static IPW is a standard static IPW estimator with normalized weights. Results are based on 2,000 replications.

	1000 observations			4000 observations		
	bias [1]	se [2]	size [3]	bias [4]	se [5]	size [6]
Baseline model	-0.005	0.044	0.065	0.001	0.011	0.061
Unobserved correlation, $\beta_u=2$	-0.003	0.049	0.058	0.001	0.012	0.048
Treatment rate, α_s =-2	-0.005	0.027	0.046	-0.003	0.007	0.049
Treatment selection, eta_s =-2	-0.000	0.047	0.054	-0.001	0.012	0.065
Treatment effect, $\delta_s = 5$	-0.000	0.047	0.054	-0.001	0.012	0.065

Table B.1: Simulated bias, size and variance of the dynamic IPW estimator

Note: Model with no treatment effect. Full generating processes described in the text. IPW estimates with bootstrapped standard errors (99 replications) and assuming true scores, respectively. Size is for 5% level tests. The results are based on 2000 replications.